NBER WORKING PAPER SERIES

IMPACT OF RURAL AND URBAN HOSPITAL CLOSURES ON INPATIENT MORTALITY

Kritee Gujral Anirban Basu

Working Paper 26182 http://www.nber.org/papers/w26182

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 August 2019

The views expressed herein are those of the authors and do not necessarily reflect the views of the National Bureau of Economic Research.

NBER working papers are circulated for discussion and comment purposes. They have not been peer-reviewed or been subject to the review by the NBER Board of Directors that accompanies official NBER publications.

© 2019 by Kritee Gujral and Anirban Basu. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

Impact of Rural and Urban Hospital Closures on Inpatient Mortality Kritee Gujral and Anirban Basu NBER Working Paper No. 26182 August 2019 JEL No. I11,I12,I14,I18

ABSTRACT

This paper examines the impact of California's hospital closures occurring from 1995-2011 on adjusted inpatient mortality for time-sensitive conditions: sepsis, stroke, asthma/chronic obstructive pulmonary disease (COPD) and acute myocardial infarction (AMI). Using a difference- in-difference approach on California's Office of Statewide Health Planning and Development (OSHPD) data, the impact of hospital closures on inpatient mortality is estimated. Outcomes of admissions in hospital service areas (HSAs) with and without closure(s) are compared before and after the closure year. The paper aims to fill gaps in prior work by using a reconciled list of California's hospital closures and by studying differential impacts of rural and urban hospital closures. To our best knowledge, this is also the first paper explicitly studying patient outcomes of California's rural closures. Results suggest that when treatment groups are not differentiated by hospital rurality, closures appear to have no measurable impact. However, estimating differential impacts of rural and urban closures shows that rural closures increase inpatient mortality by 0.46% points (an increase of 5.9%), whereas urban closures have no impact. Results differ across diagnostic conditions; the general effect of closures is to increase mortality for stroke patients by 3.1% and for AMI patients by 4.5%, and decrease mortality for asthma/COPD patients by 8.8%.

Kritee Gujral The CHOICE Institute University of Washington 1959 NE Pacific St Box 357630 Seattle, WA 98195 kritee@uw.edu

Anirban Basu The CHOICE Institute Departments of Pharmacy, Health Services, and Economics University of Washington, Seattle 1959 NE Pacific St., Box - 357660 Seattle, WA 98195 and NBER basua@uw.edu

Impact of Rural and Urban Hospital Closures on Inpatient Mortality

Kritee Gujral, Anirban Basu

July 22, 2019

Abstract

This paper examines the impact of California's hospital closures occurring from 1995-2011 on adjusted inpatient mortality for time-sensitive conditions: sepsis, stroke, asthma/chronic obstructive pulmonary disease (COPD) and acute myocardial infarction (AMI). Using a differencein-difference approach on California's Office of Statewide Health Planning and Development (OSHPD) data, the impact of hospital closures on inpatient mortality is estimated. Outcomes of admissions in hospital service areas (HSAs) with and without closure(s) are compared before and after the closure year. The paper aims to fill gaps in prior work by using a reconciled list of California's hospital closures and by studying *differential* impacts of rural and urban hospital closures. To our best knowledge, this is also the first paper explicitly studying patient outcomes of California's rural closures. Results suggest that when treatment groups are not differentiated by hospital rurality, closures appear to have no measurable impact. However, estimating differential impacts of rural and urban closures shows that rural closures increase inpatient mortality by 0.46% points (an increase of 5.9%), whereas urban closures have no impact. Results differ across diagnostic conditions; the general effect of closures is to increase mortality for stroke patients by 3.1% and for AMI patients by 4.5%, and decrease mortality for asthma/COPD patients by 8.8%.

Keywords: hospital closures, rural closures, access to care, California's hospital closures, California's rural hospitals, inpatient mortality, heart attack, stroke, asthma/COPD, sepsis

JEL Codes: I11, I12, I14, I18

1 Introduction

About 15% of U.S. hospitals have closed since 1990 and rates of hospital closure have increased over the last decade (Carroll, 2018). There is sufficient consensus pointing to weakened hospital finances owing to low reimbursement rates and uncompensated care for uninsured patients as the primary cause for hospital (and emergency department (ED)) closures (Lindrooth et al., 2018; Hodgson et al., 2015).¹ These problems are exacerbated in rural areas, especially due to the recent economic downturns and job losses leading to outmigration and shrinking populations, leaving behind populations that are older, sicker, and more reliant on Medicaid and Medicare. Such shifts lead to financially less favorable payer-mixes at hospitals, making hospitals more susceptible to closures. Other factors influencing rural closures include aging facilities, outdated payment

¹However, there is a lack of consensus regarding the link between Medicaid expansion and hospital (or ED) closures (Lindrooth et al., 2018; Allen, 2017; Friedman et al., 2016; Wishner et al., 2016). As this paper is a state-specific study, explicit discussion of this link between Medicaid expansion and hospital closures is omitted here.

and delivery system models, greater competition from urban hospitals (for patients and for federal resources), and strategic decisions by corporate owners or operators (Kaufman et al., 2016; Wishner et al., 2016; Premkumar et al., 2016).

Hospital closures can severely fracture patient access to care (especially emergency care (Wishner et al., 2016)), leading to treatment delay and adverse health outcomes (Carroll, 2018). At the same time, closures may reallocate patients to more efficient hospitals, improving patient welfare. The net impact of closures, whether rural or urban, on patient outcomes is not well-known, and remains an empirical question.

Rural hospital closures have been of particularly high concern due to increasing rates of rural closures since 2010, and because of the complexities they pose in terms of access and efficiency (Carroll, 2018; Holmes et al., 2017; Kaufman et al., 2016; Wishner et al., 2016; Escarce and Kapur, 2009). From an efficiency standpoint, rural hospitals exiting the market seems inevitable as these hospitals tend to be too small to operate efficiently (Song and Saghafian, 2019; Carroll, 2018; Hannan et al., 2003; Dranove, 1998). However, such exits can have enormous negative implications for patient welfare. Rural closures increase travel times for patients (Troske and Davis, 2019; Hsia and Shen, 2011) and lead to outmigration of health care professionals post-closure (Manlove and Whitacre, 2017; Wishner et al., 2016), severely dismembering patient access to care and exacerbating social disparities in health outcomes (Song and Saghafian, 2019). These concerns have led recent public policy efforts to devote substantial resources towards forestalling rural hospital closure Carroll (2018).

As policymakers grapple to identify instruments capable of addressing the threats hospital closures pose, gaps in understanding must also be closed. Despite the increase in hospital closure rates and prominent media calls to attention (The Takeaway, 2019; Carroll, 2019; Frakt, 2019b; Kaufman et al., 2016), very few studies have analyzed the impact of hospital closures on patient outcomes. Previous studies have instead looked at the financial and economic impact of hospital closures (Manlove and Whitacre, 2017; Hodgson et al., 2015), factors influencing hospital and ED closures (Lindrooth et al., 2018; Wishner et al., 2016; Friedman et al., 2016), and the impact of ED closures on patient outcomes (Hsia et al., 2012). Literature studying the effect of hospital closures, both rural and urban, on patient outcomes is sparse.

Existing studies of the impact of hospital (and ED) closures on patient outcomes yield mixed results and many focus exclusively on Medicare patients (Song and Saghafian, 2019; Carroll, 2018; Joynt et al., 2015; Rosenbach and Dayhoff, 1995). Song and Saghafian (2019) find that hospital closures are associated with an increase in 30-day patient mortality at hospitals, likely due to "speed-up behavior" in which hospitals reduce average length-of-stay, rather than average bed idle time. Carroll (2018) finds that patients losing their closest hospital (compared to those losing their second closest hospital) experience increased mortality. Buchmueller et al. (2006) find that hospital closures increase deaths from heart attacks and unintentional injuries in Los Angeles (LA) county. However, several studies find no impact of hospital (or ED) closures on patient mortality (Joynt et al., 2015; Hsia et al., 2012; Rosenbach and Dayhoff, 1995). Thus, whether it is studies exclusively focusing on Medicare patients (Song and Saghafian, 2019; Carroll, 2018; Joynt et al., 2015; Rosenbach and Dayhoff, 1995), or California-specific studies (Buchmueller et al., 2006; Hsia et al., 2012; Liu et al., 2014), a lack of consensus regarding the impact of hospital (and ED) closures prevails, even when studying similar geographic cohorts and diagnostic conditions in similar time periods. Furthermore, Medicare population as a whole tends to be relatively well-insured, and postclosure barriers to access may be larger for the uninsured and those covered by Medicaid (Rosenbach and Dayhoff, 1995). Nevertheless, these populations have not been sufficiently examined in the hospital closure literature.² The lack of relevant studies and uncertainty across the existing ones is crippling from a policy perspective, and could prove detrimental for patients. This warrants further scrutiny of the impact of hospital closures on patient outcomes.

This paper studies the impact of both rural and urban hospital closures in California, 1995-2011, on inpatient mortality. The analysis includes all Medicare and non-Medicare patient admissions diagnosed with sepsis, stroke, asthma/COPD, or acute myocardial infarction (AMI). These diagnostic conditions were chosen because post-closure mechanisms (such as longer travel time or overcrowding at surrounding hospitals) are likely to result in larger adverse impacts for patients with time-sensitive conditions, and these time-sensitive conditions offer more comparability with prior hospital closure studies (Carroll, 2018; Liu et al., 2014; Joynt et al., 2015; Hsia et al., 2012; Buchmueller et al., 2006). The paper is novel in two primary ways; it 1) explicitly differentiates between rural and urban closures as forms of "treatment" expected to affect patients (see motivation for this approach in Section 2.3), and 2) uses a list of California's hospital closures, which attempts to reconcile discrepancies identified in previous lists of California's closures. California is an important state for studying hospital closures, as it had an accelerated rate and the highest number of closures in the year 2000 (Nicholas C Petris Center, 2001; OIG, 2000), and remains a state with one of the highest number of closures in recent years (Troske and Davis, 2019). To our best knowledge, this is the first paper drawing explicit focus to rural closures in California. While the generalizability of state-specific studies can be limited, general insights from a data-rich state, bifurcated by rural and urban closures, can be instrumental in providing impetus for research and policy efforts in relatively more rural states that may be more vulnerable but offer lower data access.

The paper is organized as follows: Section 2 outlines the methodology; Section 2.2 presents the more common methodology estimating the impact of a general hospital closure, whereas Section 2.3 outlines the paper's novel methodology, distinguishing the impact of a rural closure vs. urban closure; Section 3 describes the data, Section 4 presents results, and Section 5 presents conclusions.

2 Methodology

Using a difference-in-difference (DID) approach, two primary specifications were used for examining the impact of hospital closures on inpatient mortality:

- General impact of hospital closures
- Differential impacts of rural and urban closures

Section 2.1 outlines the definition of "affected patients" used for the analysis, while Sections 2.2 and 2.3 detail the two above-mentioned specifications. Separate subgroup analyses were also conducted for each of the time-sensitive conditions: sepsis, stroke, asthma/COPD, and AMI.

2.1 Definition of "Affected Patients"

Several prior studies (Carroll, 2018; Hsia et al., 2012; Buchmueller et al., 2006) use change in geographic distance or driving time post-hospital closure as the primary predictor (or as the basis for

²Liu et al. (2014) and Hsia et al. (2012) do study patient outcomes for these populations in the context of ED closures. However, the impact of ED closures can differ from that of hospital closures. As some hospitals may have multiple EDs, ED closures may have a smaller or a different type of disruption of access than a hospital closure does. For instance, the loss of a full hospital is likely to create greater service interruptions than an ED closure would (Buchmueller et al., 2006). The loss of a hospital would also necessitate patients to seek care at a different hospital, whereas patients may end up at the same hospitals post-ED closures.

the primary predictor) affecting patient outcomes. In these studies, affected patients are typically those that experience a change in distance to their nearest hospital as a result of a hospital closure. While distance is an important factor determining patient choice, especially in rural areas, distance sensitivity can vary for different types of procedures (Premkumar et al., 2016) and other factors such as hospital size, technologies available, hospital quality may additionally influence hospital choice (Premkumar et al., 2016; Hodgson et al., 2015; Escarce and Kapur, 2009). Moreover, geodetic distances, driving distances, and driving times may sometimes suffer from measurement inaccuracies (Hsia et al., 2012). Given that prior California-specific studies (two of which use such metrics) (Hsia et al., 2012; Liu et al., 2014; Buchmueller et al., 2006) find differing implications for mortality (albeit due to varying aspects of ED and hospital closures), this paper takes an alternate approach.

Hospital Service Areas (HSAs), which represent local health care markets representing patients' primary care travel pattern, are used for identifying affected patients in this paper, similar to Joynt et al. (2015) and Rosenbach and Dayhoff (1995). A patient is considered affected by a hospital closure if his/her residential zip code was part of an HSA that experienced a hospital closure.

Advantages of such HSA-based identification of affected patients are:

- Implicit inclusion of the distance mechanism while avoiding the assumption that patient hospital choices/travel patterns are based strictly on distance;
- Inclusion of multiple other mechanisms:
 - Immediate detrimental outcomes, as well as lagged outcomes resulting from delayed or foregone care post-closures.
 - Ripple effects, such as overcrowding and overburdening of hospitals, on surrounding communities within the HSA (similar aim as Song and Saghafian (2019); Hodgson et al. (2015); Liu et al. (2014)). Note that overcrowding and overburdening may also lead to negative outcomes at hospitals *outside* the affected HSA. Such spillover effects on patients residing in affected HSAs that travel outside their HSA to seek care post-closure would be measured in the treatment effect. However, spillover effects for patients residing outside the affected HSA would be part of the control group effect. Thus, any adverse treatment effect due to spillovers would be conservatively estimated.
 - Potentially reduced quality at surrounding hospitals due to lower competition when prices are market determined (Frakt, 2019a).
 - Increased efficiency at surrounding hospitals (Gaynor et al., 2015).
 - Effects on patients seeking care both within and outside their residential HSA, and those switching these patterns post-closure. Note that this approach can also overcome a limitation of HSA-based identification of affected *hospitals* rather than of affected patients noted in Liu et al. (2014). That is, if affected patients switch HSAs post-closures, the effect on them, and any effect of overburdening because of them, would be missed if studying affected *hospitals*. This is especially true for rural patients, as they are likely to bypass their residential HSAs post-hospital (or ED) closures in their HSAs.
- Reduction in confounding owing to inadequate control of simultaneous mechanisms. Such confounding can occur when studying only one channel of hospital closure's impact without controlling for others. This type of confounding is mitigated here as all mechanisms are combined into one observed variable.

The primary disadvantage of such HSA-based identification of affected patients is thus the inability to distinguish between various mechanisms of impact, limiting immediate and specific

policy implications to emerge. Nonetheless, the paper studies an *overall* impact of hospital closures, in case the adverse effects are sparsely distributed across several channels of impact. Given that scholars are currently split on whether or not hospital closures have any adverse impact on patient outcomes, the paper's approach can reduce this divide to further policy efforts.

2.2 General Impact of Hospital Closures

A general effect of hospital closures is estimated first. In this portion of the analysis, no distinction is made between patients who experience a rural closure vs. those who experience an urban closure in their respective residential HSAs. Variation in timing and reasons of hospital closures throughout California from 1995 to 2011 is exploited using a "staggered" DID or DID with multiple time periods (Stevenson and Wolfers, 2006). This study design is useful when the same policy is implemented in different regions nonconcurrently, or "staggeringly." It measures the effect of a policy, hospital closure in this case, which is constant across time, reasons of closure, and geographical units.

The model is specified as follows:

$$Y_{iqt} = \alpha_0 + \alpha_1(Treat_q) + \alpha_2(TreatPost_{qt}) + \alpha_3 X_{iqt} + H_i + T_t$$
(1)

 Y_{igt} is a binary variable indicating whether an admission resulted in death at the hospital. Thus, the following probability is estimated:

$$P(Y = 1|g, t) = F(\alpha_0 + \alpha_1(Treat_g) + \alpha_2(TreatPost_{gt}) + \alpha_3X_{igt} + H_i + T_t)$$

$$(2)$$

where F is a logit function and $0 \le F \le 1$. $Treat_g$ indicates whether or not the admitted patient's HSA was ever affected in the 1995-2011 period. $TreatPost_{gt}$ indicates whether the admission is from an affected HSA and occurs in that HSA's post-closure period. X_{igt} are admission-level and hospital-level covariates that can impact inpatient mortality. H_i are hospital fixed effects and T_t are year fixed effects. α_2 is the parameter of interest.

2.3 Differential Impacts of Rural and Urban Hospital Closures

Previous studies of the impact of hospital closures on mortality treat all hospital closures alike (similar to the approach described in Section 2.2), and do not distinguish between rural and urban closures. However, there are well-documented differences in care-seeking behavior across rural and urban areas. For instance, rural patients typically experience larger travel distances, longer transportation times, and other transportation barriers, compared to urban patients (Lindrooth et al., 2018; Kaufman et al., 2016; Premkumar et al., 2016; Hsia and Shen, 2011). Rural patients are known to bypass nearby rural hospitals for nearby urban hospitals. Rural patients, compared to urban patients, are also much more likely to use hospitals for primary care needs (Premkumar et al., 2016; Wishner et al., 2016). These baseline differences in care-seeking behavior of rural and urban patients can lead to differential effects for hospital closures in rural areas vs. those in urban areas.

Using national data for the years 2011-2014, Troske and Davis (2019) demonstrate that in the aftermath of hospital closures within their zipcodes, rural patients experience a 76% increase in ambulance transportation times, compared to before closure. Although Troske and Davis (2019) find that urban patients experience no change in travel times post-closures occurring within their (urban) zip codes, urban patients residing closer to *rural* closures may nonetheless experience longer

ambulance transportation times due to shared resources with rural patients. Such spillover effects from a closure in one zipcode to surrounding zipcodes are relatively under-explored, even though half of the studies that find a negative impact of hospital (or ED) closures on mortality suggest that overcrowding of surrounding hospitals post-closures is a strong contributor towards increased mortality (Song and Saghafian, 2019; Liu et al., 2014).

Additionally, while urban patients seem to shift their usual source of care post-closure from hospitals to physicians' offices or community clinics, which are viewed as more appropriate sources of primary care (Buchmueller et al., 2006), no such favorable substitution is observed for rural patients (Rosenbach and Dayhoff, 1995). Furthermore, there may be a potential decline in physician availability in rural areas due to physicians relocating post-closure (Rosenbach and Dayhoff, 1995), exacerbating rural patients' baseline pattern of seeking primary care at hospitals.

Given that hospital closures negatively impact health care access for rural patients compared to urban patients, and given the existence of strong spillover effects post-closures, spillover effects from rural closures may be larger than those from urban closures. As such, the overall effect of a rural hospital closure, which includes transportation delays and overburdening mechanisms, is hypothesized to be larger than that of an urban closure. Recall that post-closure increases in efficiency leading to quality gains and better outcomes for patients may additionally be at play (Joynt et al., 2015) and may differ depending on whether the closing hospital was rural or urban. Hence, this paper examines the differential impacts of rural and urban closures.

There is an added advantage to this approach. Several prior studies ignore urban closures when measuring the impact of rural closures. Neglecting to account for concurrent urban closures when studying rural closures (or vice versa) confounds estimates. For instance, if urban closures decrease mortality, but are not controlled for when studying the impact of rural closures, the effect of rural closures may be overestimated. If urban closures increase mortality but are not controlled for, the impact of rural closures would be underestimated. This paper limits confounding of this sort by appropriately accounting for both types of closures.

Thus, to allow for potentially different effects of rural vs. urban closures and to control for one type of closure while examining the impact of the other, a staggered DID approach with two treatment groups is used: 1) admissions affected due to a rural hospital closure, and 2) admissions affected due to an urban hospital closure.

The model is specified as follows:

$$P(Y = 1|g, t) = F(\gamma_0 + \gamma_1(RuralTreat_g) + \gamma_2(RuralTreatPost_{gt}) + \gamma_3(UrbanTreat_g) + \gamma_4(UrbanTreatPost_{gt}) + \gamma_5 X_{igt} + H_i + T_t)$$
(3)

Note that Equation 3 above is similar to Equation 2, except in its treatment variables. $RuralTreat_g$ indicates whether or not the admitted patient's HSA was ever affected by a rural hospital closure in the 1995-2011 period. $RuralTreatPost_{gt}$ indicates whether the admission is from such an HSA and occurs in that HSA's post-rural closure period. $UrbanTreat_g$ indicates whether or not the admitted patient's HSA was ever affected by an urban hospital closure in the 1995-2011 period. $UrbanTreatPost_{gt}$ indicates whether the admission is from such an HSA and occurs in that HSA's post-rural closure period. $UrbanTreatPost_{gt}$ indicates whether the admission is from such an HSA and occurs in that HSA's post-rural closure period. $UrbanTreatPost_{gt}$ indicates whether the admission is from such an HSA and occurs in that HSA's post-rurban closure period. γ_2 and γ_4 are the parameters of interest in this specification.

2.4 Exogeneity of Treatment

Concerns may arise that the impact of hospital closure is endogenous to hospital behavior or hospital quality (Carroll, 2018). That is, hospitals of lower quality or hospitals with worse behavior

may be more likely to close. In such cases, HSAs affected by closure may exhibit worse outcomes due to worse hospitals in those areas, and not because of closure. To address such concerns, note firstly that variation in timing of closures makes it highly unlikely that results are due to changes in hospital quality or behavior (unrelated to closures) that coincide with identified closure years of the various hospitals. Additionally, hospital fixed effects are included for controlling any time-invariant hospital effects.³

Nevertheless, there may be concerns that hospital closures occur in HSAs where individual patients are relatively older, sicker, poorer, more likely to be racial minorities, and more likely to be uninsured. Thus, to control for disease severity and poverty, patient characteristics such as age, co-morbidities, admission source variables, insurance status and an indicator for whether an admission is from a rural or urban zip code are used.

3 Data

3.1 Closed Hospitals

Although a few recent studies have pointed to the lack of a centralized list of hospital closures and some data issues in identifying closures (Song and Saghafian, 2019; Carroll, 2018), none shed sufficient light on the nature and extent of these issues. This section describes the list of closed hospitals used, and compares it with other reports of California's closures, with the aim of reducing future discrepancies in the hospital closure literature.

OSHPD patient discharge data from 1995 to 2011 is used for identifying hospital closures with a process comparable to that used in Song and Saghafian (2019). A hospital is defined as "closed" if its general acute care admissions dropped to 10 or less in a given year (and remained so in subsequent years) from 1995 to $2011.^4$ "Closure year" was defined as the year *prior* to the year in which hospital admissions dropped to 10 or less.⁵

To distinguish appropriately between closed hospitals and hospitals that may exhibit 10 or fewer admissions due to a name/hospital identifier change (owing to a merger or ownership change not resulting in closure), additional sources were consulted for confirming closures. Sources consulted include government reports (OIG, 1995, 1996, 1997, 1998, 1999), closed hospitals list from the LA Times (LA Times Data Desk, 2019), and academic publications and reports (Hodgson et al., 2015; Buchmueller et al., 2006; Nicholas C Petris Center, 2001).^{6,7} After the confirmation process, a list of 139 potential closures identified using admissions data was reduced to a list of 99 hospital closures.⁸

³Note also that there is no evidence that affected HSAs or closing hospitals provide lower quality of care (even though patients may *perceive* certain rural hospitals to be of poor quality based on anecdotal evidence (Wishner et al., 2016; Escarce and Kapur, 2009)); instead, the bulk of the literature points to differences in financial health and efficiency of treated vs. non-treated facilities (a point also noted by (Carroll, 2018)).

⁴Additional sources mentioned in the next paragraph are used for identifying closures in 2011, as admissions data for 2012 were not available and thus a potential drop in admissions in 2012 cannot be identified.

⁵Alternate analyses using a modified definition of "closure year" were also conducted and are presented in Appendix C. Instead of using the year *prior*, "closure year" was defined as the very year in which a hospital's admissions dropped to 10 or less, and results of this analysis were very similar to the main results presented in the paper.

⁶Closed hospitals data, not used in the primary publication, was also obtained from authors of Hodgson et al. (2015).

⁷Numerous other sources, such as non-profit organization reports (CHCF, 2015; HASC, 2010) and news articles announcing individual hospital closures (or suggesting lack thereof), were also used, but are not all listed here. A complete list of these sources can be provided upon request.

⁸Note that this confirmation process may have resulted in excluding real but under-publicized closures, and thus remains a conservative list of closures. Nevertheless, as will be discussed later in this section, this paper ensures inclusion of the largest number of closures in California, and potential exclusion of a few such hospitals is unlikely to

Seven of the 99 hospitals, which closed and re-opened in the same year and in the same zip code, were excluded from the list of closed hospitals. The resulting 92 hospitals are the confirmed closures used for this analysis.⁹

Next, using 2004 data from United States Department of Agriculture (USDA) Economic Research Service (ERS), hospitals were assigned a rural-urban commuting area (RUCA) code based on the hospital's zip code. RUCA codes were used for classifying hospitals as rural or urban (consistent with (Holmes et al., 2017)). Rural classifications were further confirmed using other sources where hospital closures were reported and using additional internet search.¹⁰ The 92 hospitals were thus classified as follows: 16 rural and 76 urban.

Some discrepancies across other sources of hospital closure lists are worth mentioning here. Hospitals appearing in one data source for a given time period were sometimes entirely excluded from other sources reporting closures in the same period. Additionally, there were some discrepancies regarding the exact year of closure, where the margin of error was typically a year, but in a few cases, this error margin ranged from 3 to 9 years. Because of these discrepancies, the present method for identifying closures has been explicitly outlined, and the resulting list is compared to the primary sources consulted.

Table 1 compares the number of closures in each year with those reported in four of the primary sources consulted. As can be seen, there are discrepancies in number of closures per year across the sources consulted. Nevertheless, the closure list by year is largely consistent with that of the LA times. It also picks up the largest or second largest number of hospital closures in most years. However, as closure year definitions across sources vary slightly, lists by year appear less compatible than comparisons of exact hospitals in each list. Thus, Table 2 compares the actual hospitals in the closure list to each of the four primary sources consulted, disregarding the year of closure for this exercise. 77%-91% of hospital closures reported by other sources are included in this analysis. The remaining 29 excluded hospitals (i.e. 9%-13% of unmatched hospitals implicit in Table 2) are listed in Table 3 alongside the reason for their exclusion.

Direct comparisons of California's closed hospitals, which are part of national studies, were not possible, as California-specific hospital counts or names were not reported in Song and Saghafian (2019); Carroll (2018); Joynt et al. (2015), and the study period does not overlap with that in Troske and Davis (2019). However, the reliance of some of these studies (Carroll, 2018; Joynt et al., 2015) on "Lanscape Change in U.S. hospitals" from American Hospital Association (AHA, 2019) appears problematic as AHA (2019) captures very few closed hospitals from California. For instance, currently available fiscal year reports AHA (2019) indicate zero closures in California from 2006-2011, compared to 13 closures identified from 2006-2011 in the present paper. Additionally, whether a list such as AHA (2019), which includes fewer hospitals, is used during the first step of the closure identification process (Joynt et al., 2015) or only during the verification step (Carroll, 2018) can matter. That is, using such a list with fewer closures as the primary source, or as a first step, implies starting from a narrower list, resulting in an overall smaller list of closures. Whereas, if such a source is one of several sources used for verification, the risk of underestimating the ultimate number of closures is reduced.

Note that the methodology used in this paper for identifying closures through admissions data picks up most missed closures across the various lists, providing a more reconciled and complete list of closures in California from 1995 to 2011. An added advantage of this approach is the reliance on year-to-year comparisons of admissions data vs. on mere year-to-year comparisons

be a problem.

⁹A complete list of these closures is provided in supplemental materials.

¹⁰Only one hospital, which was classified as urban using RUCA codes, was updated to being "rural" based on other reporting.

of hospital listings (which do not include admissions information). The latter approach may be more susceptible to data reporting issues, as it is easier to wrongly include or exclude one hospital identifier in a given year than it is to wrongly include or exclude several admissions for a given hospital and year.

A few factors contributing to "closure year" discrepancies are also noted here. When a hospital retains its license, but has ceased to admit new general acute care patients, reported closure years may be later than years in which access to care was actually interrupted (also pointed out by Nicholas C Petris Center (2001)). When parts of the hospital remain open (e.g. ambulatory care or EDs), or when hospitals convert to entities providing outpatient or specialty care, official closure years reported may be later or different than years in which general acute care access was interrupted. Definitions of "closure year" should thus be explicitly noted in literature, and appropriate definitions must be adopted depending on the purpose of analysis. The "closure year" definition in this paper is well-aligned with the paper's intention to capture the very first instances of interruptions in general acute care. In cases of closure year discrepancies, closure year in this paper is always earlier than in other reports. Thus, if a hospital stopped providing general acute care services, but was still providing some or other aspects of care, the reduction or interruption in care due to closure, i.e. "treatment", would be smaller than if all services were completely shut off. This implies that the paper's estimates are conservative (similar to (Troske and Davis, 2019)).

Voar	No. of Closuros		Compare to of	ther sources	
rear	No. of Closures	Petris.org ¹¹	$OIG reports^{12}$	$LA times^{13}$	$\mathrm{Hodgson}^{14}$
1995	6	1	2	-	-
1996	9	3	7	-	-
1997	13	3	6	-	-
1998	10	7	10	14	1
1999	9	3	5	6	0
2000	8	6	7	12	0
2001	1	-	-	4	8
2002	3	-	-	2	6
2003	8	-	-	10	8
2004	9	-	-	6	7
2005	3	-	-	4	4
2006	1	-	-	1	2
2007	5	-	-	2	5
2008	1	-	-	1	6
2009	4	-	-	-	8
2010	2	-	-	-	3
2011	0	-	-	-	2
Total	92	23	37	62	60

Table 1: No. of Hospital Closures by Year

The second column from the left indicates the number of closures in our analysis by year. The columns to the right of it indicate the number of closures identified by year in other sources. As neither of the other sources report closures for the entire period between 1995-2011, dashes are used to indicate missing values. Note that in some cases, a closed hospital identified by a source may occur in the following or preceding year for a different source.

¹¹Nicholas C Petris Center (2001)

¹²OIG (1995, 1996, 1997, 1998, 1999, 2000)

¹³LA Times Data Desk (2019)

 $^{^{14}}$ Hodgson et al. (2015)

Other Sources	No. of Closures	No. Matched	% Matched
Petris Center report, 1995-2000	23	21	91%
OIG reports, 1995-2000	37	33	89%
Buchmueller ¹⁵ (LA county only), $1997-2003$	15	13	87%
LA times, 1998-2008	62	53	85%
Hodgson, 1995-2011	60	46	77%

Table 2: % Closed hospitals matched with other data sources

This table reports the number and percentage of closed hospitals identified by other sources that also occur in our closed hospitals list. Note that the year of closure may differ across the matched hospitals in some cases.

 $^{^{15}}$ Buchmueller et al. (2006)

1 Thesit Hospital 1996 OIG reports Recopende within a year. 2 Lakside Hospital 1996 OIG reports Rame not found in OSHPD data. Netherent hespital Dialy 3 Watsenville Comm. Hespital 1998 OIG reports Rame not found in OSHPD data. Netherent hespital 4 Surcrest Hospital, Orange Co. 1998 OIG reports Round in OSHPD data. Network Hospital 5 SHG Specialty Hospital 1998 IA times Name not found in OSHPD data. Name search. 5 Stater Foundation, FL Cajon 1998 IA times Name not found in OSHPD data. Name search. 5 Stater Foundation, FL Cajon 1009 Diricorg. Name not found in OSHPD data. Name search. 6 Stater Foundation, FL Cajon 1.A times Name not found in OSHPD data. Name not found in OSHPD data. Name search. 11 Hoops Community Hospital 2000 LA times Name not found in OSHPD data. Not confirmed in internet search. 13 Anahein Memorial Medical Crt. 2000 LA times Name not f		Hospital Name	Year	Source	Reason for exclusion
2 Masseide Hospital 1996 OIC reports Name not found in OSHPD data. 3 Wasseide Coam. Hospital 1996 OIC reports Name not found in OSHPD data. 4 Surcrest Hospital, Coam. Hospital 1998 OIC reports Re-opened within a year. 5 Step Specialty Hospital 1998 Littines Name not found in OSHPD data. Possibly closed before 1995. 5 Stater Foundation, Norwalk 1999 Littines Name not found in OSHPD data. Not confirmed in internet search. 5 Kaiser Foundation, Norwalk 1999 Littines Name not found in OSHPD data. Not confirmed in internet search. 5 Kaiser Foundation, Norwalk 1999 Littines Name not found in OSHPD data. Not confirmed in internet search. 5 Kaiser Foundation, Norwalk 1099 Did reports Name not found in OSHPD data. Not confirmed in internet search. 10 Lot 2000 Littines Name not found in OSHPD data. Not confirmed in internet search. 11 Kaiser Foundation, Medical Cr. 2000 Littines Name not found in OSHPD data.		Tustin Hospital	1996	OIG reports	Re-opened within a year under a different name, with same hospital ID.
3 Wasawrille Comm. Respital 1938 Buchmeller Re-opened within a year. 5 Kaners Hisopital, Orange Co. 1938 Littimes Name not found in OSHPD data. <i>R</i> not confirmed in internet search. 5 Kaiser Foundation, Norwalk 1938 Littimes Name not found in OSHPD data. <i>R</i> not confirmed in internet search. 5 Kaiser Foundation, Norwalk 1939 Littimes Name not found in OSHPD data. <i>R</i> or confirmed in internet search. 6 Kaiser Foundation, Norwalk 1939 Littimes Name not found in OSHPD data. <i>R</i> or confirmed in internet search. 7 Kaiser Foundation, Respende within a year under a different name. 1000 Littimes Name not found in OSHPD data. <i>R</i> not confirmed in internet search. 1 Hoopa Community Hospital 200 Littimes Name not found in OSHPD data. <i>R</i> not confirmed in internet search. 1 Hoopa Community Hospital 200 Littimes Name not found in OSHPD data. <i>R</i> not confirmed in internet search. 1 Hoopa Community Hospital 200 Littimes Name not found in OSHPD data. <i>R</i> not confirmed in internet search. 1 Hoopa Community Hospital 200 Littimes	7	Lakeside Hospital	1996	OIG reports	Name not found in OSHPD data.
4 Suncreat Haspital, Drage Co. 1998 Burdmueller Name not found in OSHPD data. Possibly closed before 1995. 5 Loss Medanose Comm. Healthcare Dist. 1998 LA times Name not found in OSHPD data. Possibly closed before 1995. 6 SHC Specially flaspital 1998 LA times Name not found in OSHPD data. Possibly closed before 1995. 7 Kaiser Foundation, Provenkik 1999 LA times Name not found in OSHPD data. Possibly closed before 1995. 9 Kaiser Foundation, Prodom. 2000 LA times Name not found in OSHPD data. Possibly closed before 1995. 10 Long Beach Community Hospital 2000 LA times Name not found in OSHPD data. Possibly closed before 1995. 11 Hoopa Community Hospital 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 13 Anabilitation lise truthe. 2000 LA times Name not found in OSHPD data. Possibly closed before 1955. 14 Hoop Community Hospital 2000 LA times Name not found in OSHPD data. Possibly closed before 1955. 15 Kaiser Pernamente, SF Forend Camp. 2000 LA times Name not found in OSHPD dat	က	Watsonville Comm. Hospital	1998	OIG reports	Re-opened within a year.
5 Loss Medanos Comu. Healthcare Dist. 1998 LA times Name not found in OSHPD data. May exist in cloarnes list under a different name. 7 Kätser Foundation, Norvalk 1999 Buchmeller Name not found in OSHPD data. May exist in cloarnes list under a different name. 7 Kätser Foundation, Norvalk 1999 Buchmeller Name not found in OSHPD data. May exist in cloarnes list under a different name. 8 Saint Louise Mental Health Ctr 1999 Buchmeller Name not found in OSHPD data. Possibly closed before 1995. 10 Long Beeds Community Heapital 2000 Lat times Re-opened within a year under a different name, with same hospital ID. 11 Hoopa Community Medical Ctr. 2000 LA times Name not found in OSHPD data. Row confirmed in internet search. 12 Kaiser Permanente. Saint Barbarz 2001 LA times Name not found in OSHPD data. Row confirmed in internet search. 13 Analein Menorial Medical Ctr. 2000 LA times Name not found in OSHPD data. Row confirmed in internet search. 14 Rain Valley Medical Ctr. 2001 LA times Name not found in OSHPD data. Row confirmed in internet search. 13 Analein Menori	4	Suncrest Hospital, Orange Co.	1998	Buchmueller	Name not found in OSHPD data & not confirmed in internet search.
6 Kaiser Foundation, Norwalk 1939 LA times Name not found in OSHPD data, k not confirmed in internet search. 7 Kaiser Foundation, Norwalk 1939 LA times Name not found in OSHPD data, k not confirmed in internet search. 8 Saint Louise Mental Health Ctr. 2000 Pertissorg. Name not found in OSHPD data, k not confirmed in internet search. 10 Loug Beach Community Med. Ctr. 2000 Pertissorg. Name not found in OSHPD data, k not confirmed in internet search. 11 Hoopa Community Media Ctr. 2000 Pertissorg. Re-opened within a year under a different name, with same hospital ID. 12 Kaiser Permanente, SF French Camp. 2001 LA times Name not found in OSHPD data, k not confirmed in internet search. 13 Habinification Institute, Santa Barbara 2001 LA times Name not found in OSHPD data k not confirmed in internet search. 14 Ukiah Walley Walley Medical Ctr. 2001 Hodgson Name not found in OSHPD data k not confirmed in internet search. 13 Ukiah Walley Medical Ctr. 2001 Hodgson Name not found in OSHPD data k not confirmed in internet search. 14 Ukiah Valley Medical Ctr.	ŋ	Los Medanos Comm. Healthcare Dist.	1998	LA times	Name not found in OSHPD data. Possibly closed before 1995.
7 Kaiser Foundation, Norwalk 1993 Brehmueller Name not found in OSHPD data & not confirmed in internet search. 8 Saint Louise Mental Heath Ctr. 1990 Futisors Name not found in OSHPD data & not confirmed in internet search. 9 Kaiser Foundation. El Cajon 1909 Petrisors Name not found in OSHPD data & not confirmed in internet search. 10 Long Beach Community Hospital 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 11 Hoapa Community Hospital 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 12 Kaiser Permanenty, SF French Camp. 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 13 Anabiein Memorial Medical Ctr. 2001 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 13 Anabiein Memorial Heapth 2001 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 14 Ukiah Valley Medical Ctr. 2001 Hodgson No general acute care admissions matched the hospital ID. 15 Rehabilitation Internet	9	SHC Specialty Hospital	1998	LA times	Name not found in OSHPD data. May exist in closures list under a different name.
8 Saint Louise Metal Health Ctr 1999 I.A times 9 Kaiser Foundation, El Cajon Detris.org, Name not found in OSHPD data. Possibly closed before 1995. 10 Long Beach Community Mespital 2000 Petris.org, Name not found in OSHPD data. Possibly closed before 1995. 11 Hoopa Community Mespital 2000 LA times Name not found in OSHPD data. Root confirmed in internet search. 12 Kaiser Permanente, SF French Camp. 2001 LA times Name not found in OSHPD data. Root confirmed in internet search. 13 Anaheim Menorial Medical Ctr. 2001 LA times Name not found in OSHPD data. Root confirmed in internet search. 14 Wish Valley Medical Ctr. 2001 LA times Name not found in OSHPD data. Root confirmed in internet search. 15 Kaiser Permanente, SF French Camp. 2001 Hodgson No general acute care admissions matched the hospital ID & k ort confirmed in internet search. 15 Kaiser Ocum. Hospital Sant Valley Medical Ctr. 2001 Hodgson No general acute care admissions matched the hospital ID & k ort confirmed in internet search. 16 Fremout Hospital Behavioral Heath 2001 Hodgson	7	Kaiser Foundation, Norwalk	1999	Buchmueller	Name not found in OSHPD data & not confirmed in internet search.
9 Kaiser Foundation, El Cajon 2000 Peris.org, LA times Name not found in OSHPD data. Possibly closed before 1995. 10 Long Beach Community Med. Ctr. 2000 Peris.org, LA times Re-opened within a year under a different name, with same hospital ID. 11 Hoopa Community Medical Ctr. 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 12 Kaiser Permanente, SF Prendi Camp. 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 13 Anaheim Menorial Medical Ctr. 2001 LA times Name not found in OSHPD data & not confirmed in internet search. 14 Usiah Valley Medical Ctr. 2001 LA times Name not found in OSHPD data & not confirmed in internet search. 15 Rababilitation Institute. Santa Barbara 2001 Hodgson No general acute care admissions matched the hospital Dt & yer on formed in internet search. 16 Fernont Hospital 2003 Hodgson No general acute care admissions matched the hospital Dt & yer confirmed in internet search. 17 Mission Comm. Hospital 2003 Hodgson No general acute care admissions matched the hospital Dt & yer confirmed in internet search.	x	Saint Louise Mental Health Ctr	1999	LA times	Name not found in OSHPD data & not confirmed in internet search.
10 Long Beach Community Med. Ctr. 2000 Petris.org, Re-opened within a year under a different name, with same hospital ID. 11 Hoopa Community Hospital 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 12 Kaiser Permanente, SF French Camp. 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 13 Anaheim Memorial Medical Ctr. 2001 LA times Name not found in OSHPD data & not confirmed in internet search. 14 Ukiah Valey Medical Ctr. 2001 Hodgson No general acute care admissions matched the hospital ID & wort confirmed in internet search. 15 Rehabilitation Institute, Sama Barbara 2001 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 16 Fremont Hospital Behavioral Health 2001 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 17 Mission Comm. Hospital 2003 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 16 Fremont Hospital 2003 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 17 Missions	6	Kaiser Foundation, El Cajon	2000	Petris.org, LA times	Name not found in OSHPD data. Possibly closed before 1995.
11 Hoopa Community Hospital 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 12 Kaiser Peramanen, SF French Camp. 2000 LA times Name not found in OSHPD data & not confirmed in internet search. 13 Anaheim Mencial Ctr. 2001 LA times Name not found in OSHPD data & not confirmed in internet search. 14 Ukiab Valey Merical Ctr. 2001 Hodgson No general acute care admissions matched the hospital ID & year not confirmed in internet search. 15 Rehabilitation Institute, Santa Barbara 2001 Hodgson No general acute care admissions matched the hospital ID & wot confirmed in internet search. 16 Fremont Hospital 2001 Hodgson No general acute care admissions matched the hospital ID & wot confirmed in internet search. 17 Mission Comm. Hospital 2003 LA times Re-opened within 2 varial acute care admissions matched the hospital ID & wot confirmed in internet search. 18 East Valley Parkins Convalescent Hospital 2003 Hodgson No general acute care admissions matched the hospital ID. 19 Lassen Community Hospital 2003 Hodgson No general acute care admissions matched the hospital ID.	10	Long Beach Community Med. Ctr.	2000	Petris.org, OIG reports	Re-opened within a year under a different name, with same hospital ID.
12Kaiser Permanente, SF French Camp.2000LA timesName not found in OSHPD data & not confirmed in internet search.13Anaheim Mienorial Medical Ctr.2001LA timesName not found in OSHPD data & not confirmed in internet search.14Ukiah Valley Medical Ctr.2001HodgsonNo general acute care admissions matched the hospital ID & vor confirmed in internet search.15Rehabilitation Institute, Santa Barbara2001HodgsonNo general acute care admissions matched the hospital ID & vor confirmed in internet search.16Fremont Hospital Behavioral Health2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search.17Mission Comm. Hospital, San Fernando2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search.17Mission Comm. Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search.18East Valley Pavilion2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search.20Hazel Hawkins Convelsectnt Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search.21Mercycare, Mercy General Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search.22Monrovia Community Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searches denoved confirme	11	Hoopa Community Hospital	2000	LA times	Name not found in OSHPD data & not confirmed in internet search.
13 Anaheim Memorial Medical Ctr. 2001 LA times Name not found in OSHPD data & not confirmed in internet search. 14 Ukiah Valley Medical Ctr. 2001 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 15 Renant Hospital Behavioral Health 2001 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 17 Mission Comm. Hospital, San Fernando 2002 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 19 East Valley Pavilion 2002 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 19 Lascen Comm. Hospital 2003 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 201 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 202 Hadgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 203 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. 203 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search. <td>12</td> <td>Kaiser Permanente, SF French Camp.</td> <td>2000</td> <td>LA times</td> <td>Name not found in OSHPD data & not confirmed in internet search.</td>	12	Kaiser Permanente, SF French Camp.	2000	LA times	Name not found in OSHPD data & not confirmed in internet search.
14Ukiah Valley Medical Ctr.2001HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search15Rehabilitation Institute, Santa Barbara2001HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search16Firemont Hospital Behavioral Health2001HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search17Rission Comm. Hospital, San Fernando2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search18East Valley Pavilion2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search19Lascen Comm. Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search20Hazel Hawkins Convalescent Hospital2003HodgsonNo general acute care admissions matched the hospital ID.20Mercycare, Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID.20Mercycare, Mercy General Hospital2003HodgsonNo general acute care admissions matched the hospital ID.21Mercycare, Mercy General Hospital2003HodgsonNo general acute care admissions matched the hospital ID.23Little Co of Mary Transitional Care2003HodgsonNo general acute care admissions matched the hospital ID.24Baywood Co ottl (Ede Medical Ctr.)2003HodgsonNo general acute care admissions m	13	Anaheim Memorial Medical Ctr.	2001	LA times	Name not found in OSHPD data $\&$ not confirmed in internet search.
15Rehabilitation Institute, Santa Barbara2001HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search16Fremont Hospital Behavioral Health2001HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search17Mission Comm. Hospital, San Fernando2002HodgsonAdmissions in OSHPD data never dropped to 10 or below.18East Valley Pavilion2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search19Lassen Comm. Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search20Hazel Hawins Convalescent Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search21Mercycare, Mercy General Hospital2005LA timesRe-opened within 2 years under a different name, with same hospital ID.23Little Co of Mary Transitional Care Ctr.2005HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search25Kaiser Fnd Hospital2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search26Siniv Valey Hospital1D & search confirmed in internet search2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in int	14	Ukiah Valley Medical Ctr.	2001	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID & year not confirmed in internet search.
16Fremont Hospital Behavioral Health2001HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search17Mission Comm. Hospital, San Fernando2002HodgsonAdmissions in OSHPD data never dropped to 10 or below.18East Valley Pavilion2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search19Lassen Comm. Hospital2003LA timesRe-opened within a year in the same zip code.20Hazel Hawkins Convalescent Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search21Mercycare, Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search22Monrovia Community Hospital2005LA timesRe-opened within 2 years under a different name, with same hospital ID.23Little Co of Mary Transitional Care Ctr2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search23Little Co of Mary Heywood2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search25Kaiser Fnd Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet search26Simi Valley Hospital2009Hodgson <t< td=""><td>15</td><td>Rehabilitation Institute, Santa Barbara</td><td>2001</td><td>$\operatorname{Hodgson}$</td><td>No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.</td></t<>	15	Rehabilitation Institute, Santa Barbara	2001	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
17Mission Comm. Hospital, San Fernando2002HodgsonAdmissions in OSHPD data never dropped to 10 or below.18East Valley Pavilion2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl19Lassen Comm. Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl20Hazel Hawkins Convelsecent Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl21Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl23Little Co of Mary Transitional Care Ctr2005HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl23Little Co of Mary Transitional Care Ctr2006HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl25Kaiser Fnd Hosp, Carson2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl26Simi Valley Hospital, Heywood2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl27Agnews State HospitalHeywood2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl <tr< td=""><td>16</td><td>Fremont Hospital Behavioral Health</td><td>2001</td><td>$\operatorname{Hodgson}$</td><td>No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.</td></tr<>	16	Fremont Hospital Behavioral Health	2001	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
18East Valley Pavilion2002HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl19Lassen Comm. Hospital2003LA timesRe-opened within a year in the same zip code.20Hazel Hawkins Convalescent Hospital2003LA timesRe-opened within a year in the same zip code.20Hazel Hawkins Convalescent Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl21Mercycare, Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl22Monrovia Community Hospital2005LA timesRe-opened within 2 years under a different name, with same hospital ID.23Little Co of Mary Transitional Care Ctr2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl25Kaiser Fnd Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl26Simi Valley Hospital, Heywood2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl27Agnews State Hospital, Heywood2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl28Palm Drive Nursing & Rehab. Ctr.2009Hodgson <td>17</td> <td>Mission Comm. Hospital, San Fernando</td> <td>2002</td> <td>$\operatorname{Hodgson}$</td> <td>Admissions in OSHPD data never dropped to 10 or below.</td>	17	Mission Comm. Hospital, San Fernando	2002	$\operatorname{Hodgson}$	Admissions in OSHPD data never dropped to 10 or below.
19Lassen Comm. Hospital2003LA timesRe-opened within a year in the same zip code.20Hazel Hawkins Convalescent Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl21Mercycare, Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl22Monrovia Community Hospital2005LA timesRe-opened within 2 years under a different name, with same hospital ID.23Little Co of Mary Transitional Care Ctr2005HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl25Kaiser Fud Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl26Simi Valley Hospital, Heywood2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl27Agnews State Hospital2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl29Lanrel Grove Hospita	18	East Valley Pavilion	2002	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
20Hazel Hawkins Convalescent Hospital2003HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl21Mercycare, Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl22Monrovia Community Hospital2005LA timesRe-opened within 2 years under a different name, with same hospital ID.23Little Co of Mary Transitional Care Ctr2005HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl25Kaiser Fnd Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl26Simi Valley Hospital, Heywood2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl27Agnews State Hospital, Heywood2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet searcl29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet searcl29Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirme	19	Lassen Comm. Hospital	2003	LA times	Re-opened within a year in the same zip code.
21Mercycare, Mercy General Hospital2004HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl22Monrovia Community Hospital2005LA timesRe-opened within 2 years under a different name, with same hospital ID.23Little Co of Mary Transitional Care Ctr2005HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl25Kaiser Fnd Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl26Simi Valley Hospital, Heywood2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl27Agnews State HospitalEventod2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searcl28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet searcl29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet searcl29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet searcl29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet	20	Hazel Hawkins Convalescent Hospital	2003	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
 22 Monrovia Community Hospital 2005 LA times Re-opened within 2 years under a different name, with same hospital ID. 23 Little Co of Mary Transitional Care Ctr 2005 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search 24 Baywood Court (Eden Medical Ctr.) 2008 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search 25 Kaiser Fnd Hosp, Carson 208 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search 26 Simi Valley Hospital, Heywood 2009 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search 27 Agnews State Hospital 209 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet search 27 Agnews State Hospital 29 Laurel Grove Hospital 209 Hodgson No general acute care admissions matched the hospital ID & year not confirmed in internet 209 Hodgson No general acute care admissions matched the hospital ID & year not confirmed in internet 28 Palm Drive Nursing & Rehab. Ctr. 2009 Hodgson No general acute care admissions matched the hospital ID & year not confirmed in internet 	21	Mercycare, Mercy General Hospital	2004	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
 23 Little Co of Mary Transitional Care Ctr 2005 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet searcles allow Medical Ctr.) 208 Hodgson No general acute care admissions matched the hospital ID & not confirmed in internet searcles in internet internet searcles in internet internet searcles in internet searcles in internet searcles internet internet searcles internet internet internet internet internet internet internet internet internet internet	22	Monrovia Community Hospital	2005	LA times	Re-opened within 2 years under a different name, with same hospital ID.
24Baywood Court (Eden Medical Ctr.)2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc25Kaiser Fnd Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc26Simi Valley Hospital, Heywood2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc27Agnews State Hospital2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet	23	Little Co of Mary Transitional Care Ctr	2005	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
25Kaiser Fnd Hosp, Carson2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc26Simi Valley Hospital, Heywood2008HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc27Agnews State Hospital2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searc28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet	24	Baywood Court (Eden Medical Ctr.)	2008	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
26Simi Valley Hospital, Heywood2008HodgonNo general acute care admissions matched the hospital ID & not confirmed in internet searched27Agnews State Hospital2009HodgonNo general acute care admissions matched the hospital ID & not confirmed in internet searched28Palm Drive Nursing & Rehab. Ctr.2009HodgonNo general acute care admissions matched the hospital ID & year not confirmed in internet29Laurel Grove HospitalID & year not confirmed in internet	25	Kaiser Fnd Hosp, Carson	2008	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
27Agnews State Hospital2009HodgsonNo general acute care admissions matched the hospital ID & not confirmed in internet searched28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet29Laurel Grove HospitalID & year not confirmed in internet	26	Simi Valley Hospital, Heywood	2008	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
28Palm Drive Nursing & Rehab. Ctr.2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet29Laurel Grove Hospital2009HodgsonNo general acute care admissions matched the hospital ID & year not confirmed in internet	27	Agnews State Hospital	2009	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID $\&$ not confirmed in internet search.
29 Laurel Grove Hospital 2009 Hodgson No general acute care admissions matched the hospital ID & year not confirmed in internet	28	Palm Drive Nursing & Rehab. Ctr.	2009	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID & year not confirmed in internet search.
	29	Laurel Grove Hospital	2009	$\operatorname{Hodgson}$	No general acute care admissions matched the hospital ID & year not confirmed in internet search.

hospitals from the closure list are presented in the right-most column.

3.2 Admission-level Analytic File

As depicted in Figure 1 below, the final admission-level analytic file was assembled using publicly available OSHPD patient discharge data, 1995-2011, for admissions with 5-digit residential zip codes,¹⁶ OSHPD Hospital Financials Data, Dartmouth Atlas of Health Care crosswalks, USDA ERS RUCA codes 2004, and closed hospitals list (described in Section 3.1).



Figure 1: Data flow used to create primary analytic files

As adverse impacts of hospital closures entail mechanisms such as increased distance, increased travel times, and increased overcrowding and overburdening for surrounding communities, patients with time-sensitive conditions are more likely to be impacted negatively. Admissions presenting the following time-sensitive conditions are included in the study: sepsis, stroke, asthma or COPD, and AMI (see Table 4 for ICD-9 diagnostic codes used for identifying these conditions). These time-sensitive conditions, in particular, were chosen to ensure better comparability with prior studies (Carroll, 2018; Liu et al., 2014; Joynt et al., 2015; Hsia et al., 2012; Buchmueller et al., 2006).

¹⁶Patient admissions whose zip codes were masked or unavailable were dropped (which includes homeless individuals and foreigners). Additionally, admissions at hospitals and zipcodes with fewer than 2000 observations for the entire study period were dropped to ensure convergence of models with hospital fixed effects and zip code-level clustering, described in Sections 2 and 3.3.

Table 4: ICD-9 codes used for patient conditions from the Clinical Classifications Software from Agency for Healthcare Quality and Research (AHRQ) (Hsia et al., 2012)

Sepsis	003.1, 020.2, 022.3, 036.2, 038.0, 038.1, 038.10, 038.11, 038.19, 038.2,
	038.3, 038.40, 038.41, 038.42, 038.43, 038.44, 038.49, 038.8, 038.9, 054.5,
	449, 790.7
Stroke	430,431,432.0,432.1,432.9,433.01,433.11,433.21,433.31,433.81,
	433.91, 434.0, 434.00, 434.01, 434.1, 434.10, 434.11, 434.9, 434.90,
	434.91, 436
Asthma/COPD	493.00, 493.01, 493.02, 493.10, 493.11, 493.12, 493.20, 493.21, 493.22,
	493.81, 493.82, 493.90, 493.91, 493.92, 490, 491.0, 491.1, 491.2, 491.20,
	491.21, 491.22, 491.8, 491.9, 492.0, 492.8, 494.0, 494, 494.1, 496
AMI	410.0, 410.00, 410.01, 410.02, 410.1, 410.10, 410.11, 410.12, 410.2,
	410.20, 410.21, 410.22, 410.3, 410.30, 410.31, 410.32, 410.4, 410.40,
	410.41, 410.42, 410.5, 410.50, 410.51, 410.52, 410.6, 410.60, 410.61,
	410.62, 410.7, 410.70, 410.71, 410.72, 410.8, 410.80, 410.81, 410.82,
	$410.9,410.90,\ 410.91,\ 410.92$

3.3 Variables

The outcome variable, inpatient mortality, is a binary variable, created using the "disposition" variable in the OSHPD discharge data, which includes in-hospital death as one of the values indicating a patient's disposition at the end of a hospital stay. This measure likely underestimates negative outcomes due to hospital closures because it does not capture out-of-hospital mortality. For instance, patients who may have died before generating a hospital admission would not be captured by this measure (similar to (Liu et al., 2014; Hsia et al., 2012)). Additionally, as an indicator of adverse outcomes, mortality captures only the relatively worse effects; smaller effects of closures may be missed. Readmission rates or morbidity may serve as better measures (Hsia et al., 2012), but were not available. As such, the results of this paper using inpatient mortality as an outcome variable would be conservative. Cause-specific mortality (used in Buchmueller et al. (2006)) would have been preferable for strengthening the causal link of the results, but such data was also not available.

Although administrative data can be restrictive in controlling for all clinical confounders, a wide range of factors influencing inpatient mortality are controlled for. These include age category, sex, race, International Classification of Diseases, Ninth Revision (ICD-9) codes, Elixhauser comorbidities (Stagg, 2015; Elixhauser et al., 1998), expected payer category (Medicare, Medi-Cal,¹⁷ private, indigent, self-pay and other), a patient's residential rurality status, and admission source variables. Admission source variables help control for disease severity, as for instance, non-elective admissions may be more urgent or severe than elective admissions (Carroll, 2018; Hsia et al., 2012); patients re-routed from another hospital's ER (and therefore experiencing longer travel time) may be more severe than those not requiring re-route (Hsia et al., 2012).

Hospital-level control variables include hospital control type (city/county, district, investor, non-profit, state), proportion of Medi-Cal days, number of medical staff, and ratio of available bed to licensed beds. Any time-invariant hospital characteristics are controlled for using hospital fixed effects. Year fixed effects were used to control for secular trends.

Additionally, population density per HSA from the year 2000, the number of time sensitive admissions per HSA for each year, Herfindahl-Hirschman Index (HHI) calculated at the HSA-level for hospitals that never closed, and county-level unemployment rates are also included.¹⁸ Summary

¹⁷Medi-Cal is California's version of Medicaid (Hodgson et al., 2015).

¹⁸Annual county-level unemployment data was obtained from California Economic Development Department's

statistics for these variables are presented in Appendix A. For the main analysis, standard errors are clustered by zip code to account for any within-zip code correlation, but errors could not be clustered for the subgroup analyses, as models with clustered errors did not converge for subgroup analyses (similar to Hsia et al. (2012)).

4 Results

Sections 4.1 and 4.2 present descriptive results, which are followed by estimates of the effects of hospital closure in Sections 4.3 and 4.4. Recall that the impact of hospital closures on inpatient mortality is estimated using two methods; one estimates the general effect of hospital closures, not distinguishing between rural and urban hospitals (described in Section 2.2), and the other estimates differential impacts of rural and urban hospital closures (described in Section 2.3). The latter is the paper's novel contribution to the literature, and is compared with the relatively more common, former approach.

4.1 Common Trends Assumption

Firstly, treatment and control group trends are presented for HSA-level inpatient mortality (weighted by HSA population density) in the left-column graphs of Figure 2. Figures 2a and 2b correspond to the specification estimating the *general* effect of hospital closures, as described in Section 2.2. Figures 2c through 2f correspond to the specification studying rural and urban closures as differential treatments, as described in Section 2.3. As the validity of the DID approach rests on the assumption that treatment and control groups follow a similar outcome trajectory in the absence of treatment, we note that all three graphs in the left column of Figure 2 exhibit that control and treatment groups roughly follow similar trends, such that the DID approach seems appropriate.

Next, the right-hand side column of Figure 2 presents average HSA-level mortality of the treatment group, before and after treatment, where the treatment year is set to zero. Note that there are no such natural treatment years for control HSAs. Such visual evidence sets expectations of the potential impacts to be discovered (Wing et al., 2018). Figure 2b and 2d suggest potential shortterm *decreases* in inpatient mortality due to general hospital closures and urban closures, whereas Figure 2f suggests that rural hospitals potentially increase inpatient mortality. As post-closure effects appear to differ across rural closure and urban closure treatments (compare Figures 2f and 2d), this paper's novel approach, which differentiates between rural and urban closures, appears critical for understanding the patient mortality impact of closures.

Labor Force Data.





(f) Rural closures - treatment group

Figure 2: (Left) Average inpatient mortality in HSAs with closure (Treatment) vs. HSAs without closure (Control). (Right) Average inpatient mortality in Treatment HSAs only, with admission year centered at the year of closure or treatment for that HSA. Averages are weighted by zip code-level population density for each HSA from the year 2000.

4.2Baseline Characteristics in Treatment HSAs vs. Control HSAs

As is typical for DID studies, it is demonstrated that baseline covariates for treatment and control groups, especially with respect to the diagnostic conditions and co-morbidities, are well-balanced. Tables 5, 6, and 7 present proportions (for dichotomous variables), means (for non-dichotomous variables) and standardized differences (calculated using formulas presented in Austin (2011)) between each of the treatment groups and the control group for diagnostic conditions, Elixhauser co-morbidities, and other covariates, respectively.

	\hat{p} , Ctrl	\hat{p} , Treat	Std. Diff.	\hat{p} , Urban	Std. Diff.	\hat{p} , Rural	Std. Diff.
AMI	0.252	0.235	-0.039	0.232	-0.046	0.260	0.018
Stroke	0.252	0.240	-0.027	0.239	-0.030	0.240	-0.027
Asthma/COPD	0.344	0.372	0.058	0.376	0.066	0.350	0.012
Sepsis	0.152	0.152	0.000	0.153	0.002	0.149	-0.008
Observations	74,687	55,925		51,854		6,236	

Table 5: Time-sensitive conditions, Proportions (\hat{p}) & Diffs., 1995

Asterisks indicate a standardized proportion difference of greater magnitude than 0.1. There are no standardized differences greater than 0.1 for the proportion of time-sensitive admissions in treatment HSAs vs. in control HSAs.

Table 6: Elixhauser Co-morbidities, Proportions (\hat{p}) & Diffs., 1995

	\hat{p} , Ctrl	\hat{p} , Treat	Std. Diff.	\hat{p} , Urban	Std. Diff.	\hat{p} , Rural	Std. Diff.
Congestive Heart Failure	0.180	0.170	-0.026	0.169	-0.028	0.197	0.043
Cardiac Arrhythmias	0.232	0.216	-0.038	0.217	-0.035	0.217	-0.035
Valvular Disease	0.067	0.059	-0.032	0.06	-0.028	0.063	-0.016
Pulmonary Circulation Disorders	0.021	0.022	0.006	0.022	0.006	0.026	0.033
Peripheral Vascular Disorders	0.044	0.043	-0.004	0.043	-0.004	0.045	0.004
Hypertension, Uncomplicated	0.320	0.32	0.000	0.321	0.002	0.331	0.023
Paralysis	0.124	0.119	-0.015	0.121	-0.009	0.109	-0.046
Other Neurological Disorders	0.097	0.098	0.003	0.099	0.006	0.086	-0.038
Chronic Pulmonary Disease	0.437	0.461	0.048	0.463	0.052	0.46	0.046
Diabetes, Uncomplicated	0.155	0.158	0.008	0.157	0.005	0.165	0.027
Diabetes, Complicated	0.044	0.038	-0.030	0.039	-0.025	0.035	-0.046
Hypothyroidism	0.052	0.050	-0.009	0.050	-0.009	0.061	0.038
Renal Failure	0.041	0.039	-0.010	0.038	-0.015	0.047	0.029
Liver Disease	0.015	0.014	-0.008	0.015	0.000	0.013	-0.017
Peptic Ulcer Disease	0.014	0.012	-0.017	0.012	-0.017	0.013	-0.008
AIDS/HIV	0.002	0.004	0.036	0.004	0.036	0.001	-0.025
Lymphoma	0.006	0.006	0.000	0.006	0.000	0.006	0.000
Metastatic Cancer	0.017	0.017	0.000	0.017	0.000	0.017	0.000
Solid Tumor Without Metastasis	0.026	0.025	-0.006	0.025	-0.006	0.029	0.018
Rheumatoid Arthritis/Collagen Vascular	0.020	0.019	-0.007	0.018	-0.014	0.017	-0.022
Coagulopathy	0.019	0.018	-0.007	0.018	-0.007	0.016	-0.022
Obesity	0.041	0.043	0.009	0.044	0.014	0.036	-0.025
Weight Loss	0.023	0.023	0.000	0.023	0.000	0.020	-0.020
Fluid and Electrolytes Disorders	0.213	0.227	0.033	0.232	0.045	0.204	-0.022
Blood Loss Anemia	0.009	0.008	-0.010	0.008	-0.010	0.004	-0.062
Deficiency Anemia	0.013	0.014	0.008	0.014	0.008	0.013	0.000
Alcohol Abuse	0.028	0.029	0.006	0.029	0.006	0.028	0.000
Drug Abuse	0.014	0.016	0.016	0.016	0.016	0.012	-0.017
Psychoses	0.010	0.011	0.009	0.012	0.019	0.009	-0.010
Depression	0.029	0.029	0.000	0.029	0.000	0.034	0.028
Hypertension, Complicated	0.062	0.065	0.012	0.065	0.012	0.061	-0.004
Observations	74,687	55,925		51,854		6,236	

Asterisks indicate a standardized proportion difference of greater magnitude than 0.1. There are no standardized differences greater than 0.1 for the Elixhauser co-morbidities between treatment and control HSAs.

	Ctrl	Treat	Std. Diff.	Urban	Std. Diff.	Rural	Std. Diff.
In-hospital death	0.078	0.078	0.000	0.078	0.000	0.084	0.021
Race							
White	0.764	0.703	-0.138*	0.69	-0.166*	0.869	0.273^{*}
Black	0.060	0.108	0.173^{*}	0.114	0.192^{*}	0.035	-0.117^{*}
Native American/Eskimo/Aleut	0.001	0.001	0.000	0.001	0.000	0.000	-0.044
Asian/Pacific Islander	0.043	0.048	0.023	0.051	0.037	0.008	-0.223*
Other	0.041	0.037	-0.020	0.037	-0.020	0.025	-0.089
Missing/Unknown	0.091	0.103	0.040	0.107	0.053	0.061	-0.113*
Admission source of patient							
Home	0.89	0.878	-0.037	0.875	-0.046	0.905	0.049
Residential Care Facility	0.011	0.008	-0.030	0.008	-0.030	0.011	0.000
Ambulatory Surgery	0.003	0.005	0.031	0.006	0.044	0.003	0.000
Skilled Nursing/Intermediate Care	0.034	0.036	0.010	0.038	0.021	0.027	-0.040
Acute/Inpatient Hospital Care	0.053	0.055	0.008	0.056	0.013	0.043	-0.046
Other (Inpatient) Hospital Care	0.005	0.006	0.013	0.006	0.013	0.006	0.013
Prison/Jail	0.000	0.001	0.044	0.001	0.044	0.002	0.063
Other	0.003	0.010	0.087	0.011	0.096	0.001	-0.044
Admission source site							
This Hospital	0.015	0.016	0.008	0.016	0.008	0.017	0.015
Another Hospital	0.059	0.065	0.024	0.066	0.028	0.047	-0.053
Not a Hospital	0.926	0.919	-0.026	0.918	-0.029	0.934	0.031
Admission route							
This Hospital's ER	0.712	0.712	0.000	0.715	0.006	0.689	-0.050
Not an ER or Another Hospital's ER	0.287	0.288	0.002	0.285	-0.004	0.308	0.045
Payer Category							
Medicare	0.545	0.511	-0.068	0.505	-0.080	0.594	0.099
Medi-Cal	0.147	0.163	0.044	0.166	0.052	0.128	-0.055
Private Payer	0.257	0.272	0.034	0.275	0.040	0.232	-0.058
Indigent	0.013	0.02	0.054	0.021	0.061	0.015	0.017
Self-Pay	0.026	0.023	-0.019	0.023	-0.019	0.019	-0.047
Missing/Unknown	0.012	0.010	-0.019	0.01	-0.019	0.013	0.009
Age Category							
Under 1 year	0.019	0.021	0.014	0.021	0.014	0.020	0.007
1-17 years	0.074	0.083	0.033	0.085	0.040	0.061	-0.051
18-34 years	0.037	0.041	0.020	0.042	0.025	0.030	-0.038
35-64 years	0.270	0.285	0.033	0.284	0.031	0.294	0.053
65 years and over	0.600	0.569	-0.062	0.567	-0.066	0.596	-0.008
Sex							
Male	0.484	0.471	-0.026	0.470	-0.028	0.472	-0.024
Female	0.516	0.529	0.026	0.530	0.028	0.528	0.024
Hospital Control Type	0.000	0.00-	0 1 6 1 4	0.005		0.100	0.00.1*
City/County	0.032	0.067	0.161*	0.065	0.154*	0.109	0.304*
District	0.136	0.037	-0.357*	0.039	-0.348*	0.009	-0.505*
Investor	0.119	0.123	0.012	0.128	0.027	0.162	0.123*
Non-Profit	0.713	0.773	0.137*	0.768	0.125^{*}	0.720	0.015
% Medi-Cal days	0.197	0.201	0.023	0.205	0.046		-0.123*
Medical Staff	304.9	350.5	0.179*	360.3	0.215^{*}	238.8	-0.331*
Available Beds/ Licensed Beds	0.918	0.898	-0.163*	0.895	-0.185*	0.947	0.293*
H5A admissions per year/1000		0.310	0.968*	0.324	1.010*	0.219	0.485^{+}
Rural admission	0.069	0.021	-0.233*	0.003	-0.361*	0.179	0.338*
Pop. Dens. per sq. mi./1000	0.460	0.741	0.474*	0.785	U.544 ⁺	0.224	-0.700* 1.949*
$\Pi \Pi$ - ΠSA for open hospitals	0/08.2	3400.1	-1.089"	3437.1	-1.083" 0.169*	33/4.4	-1.242" 1.071*
Observations	1.(01	1.098	-0.019	1.343	-0.103	13.482	1.971
Observations	(4,687	55,925		$_{51,854}$		0,230	

Table 7: Other Variables, Proportions ($\hat{p}),$ Means ($\bar{\mu})$ & Diffs., 1995

Asterisks indicate a standardized mean or proportion difference of greater magnitude than 0.1.

Although there is no universally agreed upon criterion regarding what constitutes important imbalance between treatment and control groups, a standardized difference less than 0.1 is considered negligible (Austin, 2011). Tables 5, 6, and 7 report the standardized differences between treatment and control groups, and asterisks indicate standardized differences of 0.1 or higher. As can be seen in Tables 5 and 6, all differences between control and treatment groups with respect to diagnostic conditions and co-mobidities were negligible. Nevertheless, we noted differences of greater magnitude than 0.1 for some variables presented in Table 7. Treatment groups differ from the control group with respect to racial composition, hospital control type, medical staff supply, bed availability, rurality status of patient residence, HSA admissions per year, HSA population density, HSA hospital concentration, and county unemployment rate. General hospital closure and urban closure treatment groups include fewer Whites and more Blacks in treatment vs. control groups. For the rural closure treatment, there were fewer Asians/Pacific Islanders in the treatment group compared to the control. All treatment groups, compared to the control group, consisted of proportion of patient admissions at city/county hospitals and fewer admissions at district hospitals. Treatment groups were also more likely to be in HSAs with lower HHIs than the control group. General hospital closure and urban closure treatment groups were more likely to be at hospitals with higher medical staff, lower bed availability, higher HSA-level population density, compared to the control group, whereas the rural closure treatment group were more likely to be at hospitals with lower supply of medical staff, higher bed availability, and lower population density compared to the control group. While the urban closure treatment group had lower unemployment than the control group, the rural closure treatment group had higher unemployment. There were fewer proportions of rural admissions in the general and urban treatment closure groups, compared to control, whereas there was a greater proportion of rural admissions in the rural treatment group. These differences between control and treatment groups, in the context of hospital closure, have been previously cited in Allen (2017) (hospital ownership) Hsia and Shen (2011)(race), Joynt et al. (2015) (oversupply of medical staff), Rosenbach and Dayhoff (1995) (HSA admissions), Hodgson et al. (2015) (HHI), and Wishner et al. (2016) (unemployment). Note also that the pattern of differences across treatment and control groups can vary depending on whether the closure was rural or urban, emphasizing the need for differentiating between these two treatment groups.

For the DID estimation to be valid, such differences between treatment and control groups must be stable over time and changes in treatment exposure should not be linked to changes in the distribution of covariates (Wing et al., 2018). Differences between control and treatment HSAs noted here are consistent with several previous studies examining hospital closures using different data sets from different time periods. As the distributions of the covariates appear consistent with expectations from prior work, and as there are no differences at all in the co-morbidities, the treatment and control groups seem well-balanced.

4.3 General Impact of Hospital Closures

When no distinction is made between HSAs affected due to rural closures vs. due to urban closures, there is no measurable impact of hospital closure on inpatient mortality. Marginal effects for treatment variables are presented in Table 8.

Variable	Marginal Effect (Std. Error)
Treat Post	0.0008
	(0.0011)
Treat	-0.0014
	(0.0011)
R^2	0.147
Ν	2,736,593

Table 8: General Impact of Hospital Closures

Standard errors are reported in parentheses.

Errors are clustered by patient residential zip codes.

+ p<0.10, * p<0.05. None of the variables presented in this table were statistically significant.

This lack of a measurable general impact of hospital closures is a finding consistent with Joynt et al. (2015); Hsia et al. (2012); Rosenbach and Dayhoff (1995), which also do not find any measurable impact of hospital (or ED) closures on mortality. As mentioned earlier in Section 3.3 and as noted in Hsia et al. (2012), readmissions or morbidity may prove to be better measures for estimating adverse patient outcomes due to hospital closures. As such, the results presented would be conservative estimates of the general impact of hospital closures.

Additionally, as Hsia et al. (2012) note, a majority of patients in California experience a less than 10 minutes increase in driving time post-ED closure. If changes in driving times post-hospital closures are similar, then detecting mortality differences through such a channel would be difficult. Furthermore, positive patient outcomes from mechanisms of higher efficiency and/or higher quality of care at open hospitals post-closures, which are cited as theoretical possibilities in the literature, may be at play in offsetting potentially negative effects due to closures, leading to an overall null result.

Moreover, as these findings are based on data from California, which tends to be relatively more urban than the country (Hsia et al., 2012), the impact of urban closures may be driving the general impact of closures, potentially masking adverse impact of rural closures. As driving times, transportation barriers, and spillover effects post-rural closures may differ from effects posturban closures, the differential impact of rural and urban closures must be analyzed. Results from studying differential impacts of rural and urban closures are thus presented in Section 4.4.

4.3.1 Subgroup Analyses - General Impact of Hospital Closures

In this section, separate estimates of the general impact of hospital closures are presented for each of the four diagnostic conditions. Marginal effects for the treatment variables are present in Table 9.

Variable	Sepsis	Stroke	Asthma/COPD	AMI
Treat Post	-0.0021	0.0033^{*}	-0.0014*	0.0033*
	(0.0024)	(0.0017)	(0.0005)	(0.0014)
Treat	-0.0004	-0.0045^{*}	0.0006	-0.0008
	(0.0025)	(0.0017)	(0.0006)	(0.0014)
R^2	0.111	0.054	0.187	0.159
Ν	$724,\!182$	678,069	956,206	691,516

 Table 9: Subgroup Analyses - General Impact of Hospital Closures

Errors were not clustered by patient residential zip codes because the model did not converge with clustering. + p<0.10, * p<0.05

Similar to the results for all conditions combined, presented in Section 4.3 above, there is no measurable impact of hospital closures for sepsis patients. However, closures increase mortality for stroke patients by 3.1% and for AMI patients by 4.5%, whereas closures decrease mortality by 8.8% for Asthma/COPD patients¹⁹ This differential effect across Asthma/COPD and AMI and stroke patients is similar to differential findings across the same diagnostic categories in Buchmueller et al. (2006) (although they find a null result for COPD patients, when studying the effect of change in distances post-urban closures in LA county). As AMI and stroke are relatively more time-sensitive condition than asthma/COPD, results suggest that negative effects of closures may be stronger for more time-sensitive conditions, and that patients with other diagnostic conditions may be experiencing net effects of closures that are positive (similar to reduced mortality for patients with chronic conditions, noted in Buchmueller et al. (2006)), potentially owing to higher efficiency at open hospitals. Further examination of the underlying mechanisms and of the various diagnostic conditions is needed for targeting and improving protective policy measures undertaken post-closures.

4.4 Differential Impacts of Rural and Urban Closures

In this section, the differential impacts of rural and urban closures on inpatient mortality are presented. Marginal effects for the treatment variables for the subgroup analyses are presented in Table $10.^{20}$ It is seen that rural hospital closures increase inpatient mortality by 0.46% points (an increase of 5.9%) whereas urban closures have no measurable impact on inpatient mortality. These results are consistent with Song and Saghafian (2019); Carroll (2018); Liu et al. (2014), which suggest that hospital closures and ED closures increase patient mortality. Song and Saghafian (2019) find that hospitals experience greater demand post-closure, and respond by using a "speed-up behavior," i.e. by reducing service duration rather than bed idle time. This leads to increased efficiency, which is nonetheless accompanied by a 3% increase in mortality. Carroll (2018) finds that rural hospital closures increase in inpatient mortality of 5% for the general patient population, and experience increases of higher magnitudes for three of the four time sensitive conditions: sepsis, 8%, stroke 10%, and AMI 10%. Although direct comparisons with prior studies are not possible due to different data sets and differing approaches, the estimates of mortality due to closures are similar, ranging from 3%-10%.

¹⁹Average inpatient mortality rate for the control group with time-sensitive conditions studied were as follows: sepsis, 17.1%; stroke 10.6%; asthma/COPD 1.6%; AMI, 7.3%.

²⁰Marginal effects for all variables are presented in Table 13 in Appendix B

Variable	Marginal Effect (Std. Error)
Rural Treat Post	0.0046*
	(0.0020)
Urban Treat Post	0.0009
	(0.0011)
Rural Treat	-0.0069*
	(0.0024)
Urban Treat	-0.0013
	(0.0012)
R^2	0.147
Ν	2,736,593

Table 10: Differential Impacts of Rural and Urban Closures

Standard errors are reported in parentheses. Errors are clustered by patient residential zip codes. + p<0.10, * p<0.05

4.4.1 Subgroup Analyses - Differential Impacts of Rural and Urban Closures

Each time-sensitive condition is also separately examined. Marginal effects of treatment variables are presented in Table 11. For sepsis patients, rural closures increase inpatient mortality by 1.54% points (an increase of 9.0%).²¹ This is consistent with the 8% increase in mortality for sepsis patients due to nearby ED closures in California owing to delayed care and overcrowded EDs, noted in Liu et al. (2014).

Urban closures increase inpatient mortality for AMI patients by 0.30% points (a 4.1% increase), similar to Buchmueller et al. (2006), which suggests a 6.5% increase in cause-specific mortality due to urban closures in LA county for AMI patients. Yet, urban closures decrease inpatient mortality for asthma/COPD patients by .10% points (about 6.3% decrease). A similar differential effect of urban closures across heart attack and COPD patients is noted in Buchmueller et al. (2006), although they find a null result for COPD patients and focus on the impact of change in distance. A negative effect for AMI, which is more time-sensitive than asthma/COPD, may be suggestive of increased distance, travel time, and/or overburdened emergency care mechanisms dominating other mechanisms for conditions that are relatively more time-sensitive. Reduced patient mortality for asthma/COPD patients, in contrast, may reflect patient welfare-improving mechanisms of hospital closure being more dominant for conditions that are less time-sensitive (similar to reduction in mortality post-closure for chronic patients (Buchmueller et al., 2006)). In any case, note that combining these diagnostic conditions, for which the effects are opposing, can result in a null result when all conditions are combined, as is the case for results presented in Table 10.

Overall, the results for sepsis and AMI patients are consistent with expectations of delayed or forgone care. However, further research on each of the underlying mechanisms for the various diagnostic conditions is needed for identifying exact channels and cautionary steps to be taken post-closure.

 $^{^{21} \}text{Baseline mortality for the time-sensitive conditions studied were as follows: sepsis, 17.1%; stroke 10.6%; asthma/COPD 1.6%; AMI, 7.3%.$

Variable	Sepsis	Stroke	Asthma/COPD	AMI
Rural Treat Post	0.0154^{*}	0.0038	-0.0015	0.0009
	(0.0044)	(0.0033)	(0.0009)	(0.0026)
Urban Treat Post	-0.0025	0.0029	-0.0010+	0.0030^{*}
	(0.0026)	(0.0018)	(0.0006)	(0.0015)
Rural Treat	-0.0268*	-0.0145*	0.0001	0.0021
	(0.0050)	(0.0038)	(0.0014)	(0.0033)
Urban Treat	0.0018	-0.0039*	0.0005	-0.0009
	(0.0026)	(0.0018)	(0.0006)	(0.0015)
R^2	0.111	0.054	0.187	0.159
Ν	724,182	678,069	956,206	691,516

Table 11: Subgroup Analyses - Differential Impacts of Rural and Urban Closures

Errors were not clustered by patient residential zip codes because the model did not converge with clustering. + p<0.10, * p<0.05

4.4.2 Sensitivity Analyses

The results presented in Section 4.4 are robust to several sensitivity checks. Firstly, one may believe that the negative impact of rural closures picks up only the effect on rural patients, which may be driven by changes in trends across rural and urban patients that are not controlled for. However, this notion is ruled out, as very similar negative effects of rural closures are observed even when restricting the analysis to include only urban patients. This also points to spillover effects of rural closures leading to increased mortality for urban patients. Marginal effects for treatment variables are presented in Table 18 in Appendix D.

Secondly, as the results for subgroup analyses could not be clustered by zip codes when analyzing each condition separately, a model with clustered errors is run for sepsis, stroke, and AMI conditions combined (i.e. excluding asthma/COPD patients). As these three conditions are relatively more time-sensitive than asthma/COPD, the adverse impact of hospital closures should be larger for these conditions than all four conditions combined. As expected, the results are stronger for these conditions, i.e. rural closures increase mortality by 0.93% points (an increase of 8.0%).²² Marginal effects for treatment variables are presented in Table 19 in Appendix D.

Next, as length of stay can be correlated with quality of care, the impact of hospital closures on length of stay (LOS) is also examined. Rural closures increase LOS by 5.5 hours. For sepsis, stroke, and AMI patients combined, rural closures increase LOS by 8 hours, whereas no impact is noted when analyzing only asthma/COPD admissions - a pattern that coincides with the mortality impact of hospital closures. Marginal effects for treatment variables for both of these analyses are presented in Table 20 and 21 in Appendix D. Nevertheless, while longer LOSs are associated with lower efficiency, the link between LOSs and quality is less clear. Shorter LOSs can be indicative of poor quality of care as hospitals respond to financial pressures by reducing service durations and discharge patients earlier than is appropriate. At the same time, poor quality of care or ED crowding may lead to complications, resulting in longer LOSs. However, despite the recent trend of shortening LOSs, most recent studies (Tran et al., 2018) do not find that shorter LOSs are linked to increased mortality (although they may be linked to readmissions). Instead, there is evidence linking increased LOS with poor quality of care and mortality (McRae et al., 2017; Chaou et al., 2016; Pines et al., 2010; Thomas et al., 1997). Regardless, even if increased LOSs are not caused

 $^{^{22}\}mathrm{Average}$ inpatient mortality rate for the control group with sepsis, stroke, and AMI admissions combined was 11.6%

by poor quality of care, they can contribute significantly to overcrowding (Henneman et al., 2010). Thus, the coinciding pattern of hospital closure's impact on LOS and on mortality is suggestive of underlying overcrowding.

Finally, one may be concerned that as rural areas suffer from declining populations, the observed results are driven by shrinking populations. However, these results are robust to the inclusion of average HSA-level population density growth from 2000 to 2010. Marginal effects for treatment variables are presented in Table 22 in Appendix D.Thus, the results observed do not seem driven by changes in population density.

4.4.3 Discussion

The differential impacts of rural and urban closures are consistent with expectations, and offer one possible explanation for the difference in results across studies that find no impact of closures vs. studies that find an adverse impact. That is, rural closures may have an overall more adverse impact on patient mortality than urban closures.

It is generally well known that patients from rural areas travel longer distances to seek care, compared to urban patients. However, distance tends not to be the only issue; several transportation barriers add to delayed or forgone patient care. In case studies of hospital closures in Kansas, Kentucky, and South Carolina, respondents from all studies indicated the need to ensure emergency transportation to neighboring hospitals post-closure (Wishner et al., 2016). Many rural communities lack adequate public transportation systems, implying that patients without cars or patients that do not drive must rely on someone to drive them. As many of the elderly cannot drive, travel tends to pose greater issues for them. Additionally, patients in rural areas worry they may not have transportation to go back from the hospital. Such transportation issues in rural areas, which lead to delayed care and foregone care, are exacerbated post-hospital closures (Wishner et al., 2016). Similar access concerns for the elderly and low-income populations have been noted in the context of California's urban hospital closures and trauma center closures (Hsia and Shen, 2011; Buchmueller et al., 2006), but such concerns and perceived reduction in access to care do not seem to affect California's population at large (Hsia et al., 2012; Buchmueller et al., 2006).

As mentioned earlier, Troske and Davis (2019) find that rural patients experience an ambulance time increase of 76% (11 minutes) in the year post-closure (within their zip code), compared to before closure. Additionally, they spend more time in an ambulance, compared to urban patients facing closure, who experience no change in ambulance time post-closure (similar to small change in distances post ED-closures, noted in (Hsia et al., 2012)).

Furthermore, rural hospital closures lead to an outmigration of health care professionals, and worsen pre-existing challenges to access. For instance, when large hospital systems in Kansas, Kentucky, and South Carolina, that had employed local physicians, closed, physicians were offered competitive salaries at other hospitals in the hospital owners' systems, causing physician relocations. In some cases, physicians were banned from entering into contracts with hospitals in the closure area. Such relocation sometimes even occurred before the hospital closed. Thus, physician relocation due to hospital closure exacerbates existing recruitment difficulties and systemic workforce shortages in rural areas (Wishner et al., 2016).

Overall, rural closures negatively impact economic measures in a community, such as poverty level, income, unemployment, and home values, in both the short- and long-term. Note that rural closures lead to job loss not just in healthcare, but also in other industries and occupations (Manlove and Whitacre, 2017). Although the direct link between unemployment and health outcomes suggests an improvement in health outcomes (mostly due to healthier behaviors) during times of unemployment Strumpf et al. (2017); Ruhm (2007, 2005, 2000), hospital and ED closure literature emphasizes disproportionately worse outcomes post-closure for lower-income populations, as these populations are most affected by access barriers (Song and Saghafian, 2019; Hsia and Shen, 2011; Buchmueller et al., 2006). Thus, such economic impacts of rural closures may have long last-ing negative impacts on access to health care that lead to deteriorated patient health for these communities.

Apart from exacerbated transportation and economic problems affecting rural patients, spillover effects of hospital (and ED) closures, which appear substantial according to Song and Saghafian (2019); Liu et al. (2014), may differ for rural closures compared to urban closures. Communities surrounding rural closures are more likely to experience longer-term spillovers, spillovers from disproportionately longer ambulance transportation times, and additional overcrowding owing to patients seeking primary care at hospitals, compared to communities surrounding urban closures.

This paper is unable to pinpoint which mechanisms play a larger role in the differential results of rural and urban closures observed. Nevertheless, these results demonstrate the extent to which negative effects of rural closures are masked when no distinction is made between treatment groups. As such, these results offer a crucial potential explanation for the lack of consensus among scholars regarding the impact of hospital (and ED) closures: rural hospital closures have a differential impact on patient mortality than urban closures.

5 Conclusions

With the rise in hospital closures, and especially with increasing rates of rural closures, lack of studies on the impact of closures on patient outcomes and the lack of consensus across existing studies pose barriers to policy implementation. Among the handful of studies examining the impact on patient mortality due to hospital (and ED) closures, about half find no impact, while the remaining studies find increased mortality for certain diagnostic conditions yet no clear impact for others. Additionally, rampant discrepancies exist across most studies regarding the number (and identity) of California's closed hospitals in a given time period. This paper attempts to reduce the aforementioned gaps and reconcile differences across prior studies. Firstly, the paper explicitly outlines its methodology for identifying closures and closure years, compiling a more complete list of California's hospital closures, 1995-2011, which is compared to closure lists from other sources. The paper also suggests methodological improvements for tracing detrimental impacts of hospital closure which may have appeared elusive or nonexistent based on prior research. Finally, this is one of the very few papers that studies the impact of hospital closures on Medicare as well as non-Medicare patients, and to our best knowledge, is the first paper to examine patient mortality impact of rural closures in California. As California has historically had and continues to have a high number of hospital closures, understanding the impact of hospital closures, and not just ED closures, is critical. Despite limitations to generalizability that come with a state-specific study, the paper's primary contributions are likely applicable beyond the context of California.

To summarize, the paper finds no impact of hospital closures when all closures are considered alike. However, when studying each diagnostic condition separately, closures are found to increase mortality for stroke patients by 3.1%, for AMI patients by 4.5%, and to decrease mortality for asthma/COPD patients by 8.8%. When differentiating between rural and urban closures, rural hospital closures increase mortality by about 5.9% overall, whereas urban hospital closures have no measurable impact. When studying each diagnostic condition separately, it is seen that rural closures increase mortality for sepsis patients by 9.0%, urban closures increase mortality for AMI patients by 4.1%, and urban closures decrease mortality for asthma/COPD patients by 6.3%.

The paper emphasizes the extent to which detrimental impacts of rural closures can get masked

in an analysis that does not distinguish between rural and urban closures. Similarly, as illustrated through the subgroup analyses, combining different diagnostic conditions, which may have opposite effects, can be problematic when studying several conditions combined. Additional research that distinguishes between rural and urban closures must be conducted for various diagnostic conditions separately in order to deepen and refine the understanding of patient mortality impact of hospital closures.

As mentioned earlier, although this paper captures multiple channels through which hospital closures can impact patient mortality, it is unable to distinguish between these mechanisms. Further research is needed for understanding each individual channel of impact, keeping in mind that adverse effects on mortality may be sparsely distributed across these channels. Using an intermediate measure of adverse patient outcomes, such as readmission or co-morbidity instead of mortality, may be particularly helpful for studies attempting to isolate the impact of a given channel.

Nevertheless, regardless of the underlying mechanisms, this paper contributes to the emerging concord in very recent studies that hospital (or ED) closures do lead to increased patient mortality, ranging from about 3-10%, which may be distributed across various channels. Overburdening of the health care system post-closure and rural closures are particularly concerning. While policy instruments that tackle root causes for closures and macroeconomic effects of rural closures appear less straightforward, there seems to be a clear need to ensure emergency transportation post-closures, especially for vulnerable populations, and to identify strategies that help surrounding hospitals manage increased and potentially differential health-care demands post-closures.

A Appendix - Sample Summary Statistics

Variable	Mean	Std. Dev.	Min.	Max.
In-hospital death	0.086	0.281	0	1
Urban Treat Post	0.434	0.495	0	1
Urban Treat	0.355	0.478	0	1
Rural Treat	0.069	0.254	0	1
Rural Treat Post	0.042	0.202	0	1
Age Categories (Base=Under 1 year or missing				
1-17 years	0.049	0.216	0	1
18-34 years	0.027	0.162	0	1
35-64 years	0.306	0.46	0	1
65 years & over	0.605	0.488	0	1
Gender (Base = Male)				
Female	0.523	0.499	0	1
Other	0.000	0.001	0	1
Race (Base = White)				
Black	0.085	0.278	0	1
Native American/Eskimo/Aleut	0.000	0.030	0	1
Asian/Pacific Islander	0.060	0.238	0	1
Other	0.064	0.244	0	1
Missing	0.103	0.304	0	1
Expected Payment (Base=Medicare)				
Medi-Cal	0.152	0.359	0	1
Private Coverage	0.208	0.406	0	1
Indigent	0.017	0.130	0	1
Self-Pay	0.027	0.164	0	1
Other or Missing	0.012	0.112	0	1
Admission Source Site (Base=Home)				
Residential Care Facility	0.015	0.121	0	1
Ambulatory Surgery	0.003	0.058	0	1
Skilled Nursing/Intermediate Care	0.056	0.231	0	1
Acute/Inpatient Hospital Care	0.047	0.213	0	1
Other (Inpatient) Hospital Care	0.005	0.073	0	1
Newborn	0.000	0.004	0	1
Prison/Jail	0.000	0.030	0	1
Other	0.007	0.083	0	1
Missing	0.000	0.016	0	1
Admission Source Site License (Base=This Hospital)				
Another hospital	0.061	0.24	0	1
Not a hospital	0.927	0.259	0	1
Missing	0.000	0.015	0	1
Patient Route (Base=Admitting hospital's ER)				
No ER or another facility's ER	0.191	0.393	0	1
Missing	0.000	0.015	0	1
Hospital Control Type (Base=City/County)				
District	0.069	0.253	0	1
Investor	0.164	0.370	0	1
Non-profit	0.720	0.448	0	1
State	0.002	0.050	0	1
Proportion of Medi-Cal Days	0.216	0.173	0	0.871
No. of Medical Staff	0.461	0.349	0	2.309
Available Beds/Licensed Beds	0.941	0.102	0.093	1.670
Rural admission	0.048	0.215	0	1
County unemployment rate	7.511	3.348	2	29.8
Pop. dens. per sq. mile by 10,000	0.606	0.624	0.002	4.983

Table 12: Sample Summary Statistics

Table 12 Continued:	Sample	Summary	Statistics
---------------------	--------	---------	------------

HSA Admissions/10000	0.312	0.358	0.006	1.891
HHI-HSA for open hospitals	5618.7	3444.4	0	10000
Elixhauser Comorbidities				
Congestive Heart Failure	0.220	0.414	0	1
Cardiac Arrhythmias	0.260	0.438	0	1
Valvular Disease	0.069	0.254	0	1
Pulmonary Circulation Disorders	0.031	0.175	0	1
Peripheral Vascular Disorders	0.069	0.253	0	1
Hypertension, Uncomplicated	0.431	0.495	0	1
Paralysis	0.089	0.285	0	1
Other Neurological Disorders	0.120	0.325	0	1
Chronic Pulmonary Disease	0.447	0.497	0	1
Diabetes, Uncomplicated	0.219	0.413	0	1
Diabetes, Complicated	0.066	0.248	0	1
Hypothyroidism	0.095	0.294	0	1
Renal Failure	0.116	0.321	0	1
Liver Disease	0.034	0.183	0	1
Peptic Ulcer Disease Excluding Bleeding	0.012	0.109	0	1
AIDS/HIV	0.002	0.05	0	1
Lymphoma	0.008	0.089	0	1
Metastatic Cancer	0.020	0.14	0	1
Solid Tumor Without Metastasis	0.034	0.183	0	1
Rheumatoid Arthritis/Collagen Vascular	0.026	0.159	0	1
Coagulopathy	0.044	0.206	0	1
Obesity	0.083	0.276	0	1
Weight Loss	0.055	0.228	0	1
Fluid and Electrolyte Disorders	0.264	0.441	0	1
Blood Loss Anemia	0.009	0.098	0	1
Deficiency Anemia	0.026	0.161	0	1
Alcohol Abuse	0.036	0.186	0	1
Drug Abuse	0.026	0.16	0	1
Psychoses	0.018	0.134	0	1
Depression	0.076	0.266	0	1
Hypertension, Complicated	0.125	0.33	0	1
Admission Year (Base=1995)				
1996	0.049	0.216	0	1
1997	0.052	0.222	0	1
1998	0.053	0.225	0	1
1999	0.054	0.226	0	1
2000	0.049	0.217	0	1
2001	0.058	0.235	0	1
2002	0.058	0.234	0	1
2003	0.056	0.23	0	1
2004	0.048	0.215	0	1
2005	0.056	0.23	0	1
2006	0.06	0.239	0	1
2007	0.055	0.228	0	1
2008	0.069	0.254	0	1
2009	0.070	0.255	0	1
2010	0.077	0.267	0	1
2011	0.080	0.272	0	1
Ν	2,736,593			

B Appendix - Differential Impacts of Rural and Urban Closures- Marg. Effects for all variables

Variable	Marginal Effect (Std. Error)
Rural Treat Post	0.005*
	(0.002)
Urban Treat Post	0.001
	(0.001)
Rural Treat	-0.007*
	(0.002)
Urban Treat	-0.001
	(0.001)
Age Categories (Base=Under 1 year or missing)	
1-17 years	-0.002*
	(0.001)
18-34 years	0.030*
	(0.001)
35-64 years	0.055^{*}
v	(0.001)
65 years & over	0.090^{*}
,	(0.001)
Gender (Base = Male)	
Female	0.006^{*}
	(0.000)
Other	0.136
	(0.127)
Bace (Base = White)	(0.121)
Black	-0.018*
Dittor	(0.010)
Native American/Eskimo/Aleut	-0.031*
	(0.001)
Asian/Pacific Islander	-0.007*
	(0.001)
Other	-0.007*
Other	(0.001)
Missing	0.001)
Wilsbillg	(0.012)
Expected Payment (Base-Medicare)	(0.001)
Modi Col	0.006*
Meul-Cai	(0,001)
Privata Covaraça	(0.001)
I IIvate Coverage	-0.004
Indiant	(0.001)
maigent	-0.042
	(0.002)
Self-Pay	0.028**
5. ft - 1	(0.002)
Missing	0.006*
	(0.002)
Admission Source (Base=Home)	
Residential Care Facility	0.015*
	(0.001)
Ambulatory Surgery	-0.039*
	(0.002)
Skilled Nursing/Intermediate Care	0.050^{*}
	(0.001)

Table 13: Differential Impacts of Rural and Urban Closures - Marg. Effects for all variables

Acute/Inpatient Hospital Care	0.015^{*}
	(0.002)
Other (Inpatient) Hospital Care	0.023*
NT 1	(0.003)
Newborn	-0.025
	(0.031)
Prison/Jail	-0.008
Other	(0.007)
Other	(0.023)
Missing	0.003
missing	(0.020)
Admission Source Site License (Base= This Hospital)	(0.000)
Another hospital	-0.056*
X X	(0.003)
Not a hospital	-0.060*
	(0.003)
Missing	-0.001
	(0.051)
Patient Route (Base=Admitting hospital's ER)	
No ER or another facility's ER	-0.024*
	(0.001)
Missing	-0.057*
	(0.017)
Hospital Control Type (Base=City/County)	0.000
District	-0.008
Invector	(0.009)
liivestoi	(0.007)
Non-profit	-0.010
Ton pront	(0.008)
State	-0.009*
	(0.004)
% of Medi-Cal Days	0.006
	(0.005)
No. of Medical Staff/1000	0.004^{*}
	(0.001)
Available Beds/Licensed Beds	-0.003
	(0.003)
Rural admission	0.004*
	(0.002)
County unemployment rate (%)	-0.000
Pon dong non ag mile/10000	(0.000)
1 op. dens. per sq. mile/10000	(0.002)
HHLHSA for open hospitals/1000	-0.001
	(0.000)
HSA Admissions/10000	0.004*
	(0.001)
Elixhauser Comorbidities	× /
Congestive Heart Failure	0.028^{*}
	(0.000)
Cardiac Arrhythmias	0.037^{*}
	(0.000)
Valvular Disease	-0.010*
	(0.001)
Pulmonary Circulation Disorders	0.029^{*}

Table 13 Continued: Diff. Impacts. - Marg. Effects for all variables

	(0.001)
Peripheral Vascular Disorders	(0.001) 0.018^*
	(0.001)
Hypertension, Uncomplicated	-0.016*
	(0.000)
Paralysis	-0.005*
Other Neurological Disorders	(0.001) 0.037*
	(0.001)
Chronic Pulmonary Disease	-0.050*
	(0.001)
Diabetes, Uncomplicated	-0.005*
Diabetes Complicated	(0.000) -0.011*
Diabetes, complicated	(0.001)
Hypothyroidism	-0.008*
	(0.001)
Renal Failure	0.043*
Liver Dicease	(0.001)
Liver Disease	(0.003)
Peptic Ulcer Disease Excluding Bleeding	-0.020*
	(0.002)
AIDS/HIV	0.018*
T I	(0.003)
Lymphoma	0.041^{*}
Metastatic Cancer	(0.001) 0.060*
	(0.001)
Solid Tumor Without Metastasis	0.031^{*}
	(0.001)
Rheumatoid Arthritis/Collagen Vascular	0.007^{*}
vCoagulopathy	(0.001) 0.057^*
· · · · · · · · · · · · · · · · · · ·	(0.001)
Obesity	-0.023*
	(0.001)
Weight Loss	0.024^{*}
Fluid and Electrolyte Disorders	(0.001) 0.046*
	(0.000)
Blood Loss Anemia	-0.009*
	(0.001)
Deficiency Anemia	-0.040^{*}
Alcohol Abuse	(0.001) 0.012*
	(0.001)
Drug Abuse	0.001
	(0.001)
Psychoses	-0.028*
Depression	(0.001) -0.026*
- Chronologi	(0.001)
Hypertension, Complicated	-0.028*
	(0.001)
Admission Year (Base=1995)	0.00.1*
1990	-0.004^{*}

Table 13 Continued: Diff. Impacts. - Marg. Effects for all variables

	(0.001)
1997	-0.003*
	(0.001)
1998	-0.003*
	(0.001)
1999	-0.000
	(0.002)
2000	0.002
	(0.002)
2001	0.003*
	(0.001)
2002	0.001
	(0.001)
2003	-0.002
	(0.001)
2004	-0.001
	(0.001)
2005	-0.003*
	(0.001)
2006	-0.004*
	(0.002)
2007	-0.008*
	(0.002)
2008	-0.013*
	(0.001)
2009	-0.020*
	(0.002)
2010	-0.027*
	(0.002)
2011	-0.030*
52	(0.002)
<i>R</i> ²	0.147
N	2,736,593

Table 13 Continued: Diff. Impacts. - Marg. Effects for all variables

Standard errors are reported in parentheses. Errors are clustered by patient residential zip codes. The model also includes hospital fixed effects. + p<0.10, * p<0.05

C Appendix - Sensitivity analysis using alternate definition of "closure year"

The following tables present results using an alternate definition of "closure year." For these analyses, "closure year" was defined as the very year in which a hospital's admissions dropped to 10 or less (instead of using the year *prior*, as is the case for the analysis presented in the main text of the paper). These results are very similar to those presented in the paper.

Table 14: General In	npact of Hospital	Closures, Alternate	"closure year"
----------------------	-------------------	---------------------	----------------

Variable	Marginal Effect (Std. Error)
Treat Post	0.0013
	(0.0010)
Treat	-0.0017+
	(0.0010)
R^2	0.147
Ν	2,729,670

Standard errors are reported in parentheses.

Errors are clustered by patient residential zip codes.

+ p<0.10, * p<0.05. None of the variables presented in this table were statistically significant.

Table 15: Subgroup Analyses, General Impact of Hospital Closures, Alternate "closure year"

Variable	Sepsis	Stroke	Asthma/COPD	\mathbf{AMI}
Treat Post	-0.0043*	0.0042^{*}	-0.0007	0.0037^{*}
	(0.0022)	(0.0016)	(0.0005)	(0.0013)
Treat	0.0013	-0.0047*	0.0001	-0.0010
	(0.0023)	(0.0016)	(0.0005)	(0.0013)
R^2	0.111	0.054	0.187	0.159
N	720,254	677,323	954,339	690,201

Standard errors are reported in parentheses.

Errors were not clustered by patient residential zip codes because the model did not converge with clustering. + p < 0.10, * p < 0.05

Table 16: Differential Impacts of Rural and Urban Closures, Alternate "closure year"

Variable	Marginal Effect (Std. Error)
Rural Treat Post	0.0053*
	(0.0019)
Urban Treat Post	0.0015
	(0.0010)
Rural Treat	-0.0077*
	(0.0023)
Urban Treat	-0.0016
	(0.0011)
R^2	0.147
Ν	2,729,670

Standard errors are reported in parentheses.

Errors are clustered by patient residential zip codes.

+ p<0.10, * p<0.05

Variable	Sepsis	Stroke	Asthma/COPD	AMI
Rural Treat Post	0.0152^{*}	0.0054	-0.0008	0.0017
	(0.0042)	(0.0033)	(0.0010)	(0.0026)
Urban Treat Post	-0.0044 +	0.0037^{*}	-0.0005	0.0037^{*}
	(0.0023)	(0.0016)	(0.0005)	(0.0014)
Rural Treat	-0.0269*	-0.0149*	-0.0010	0.0005
	(0.0048)	(0.0037)	(0.0013)	(0.0032)
Urban Treat	0.0037	-0.0040*	0.0001	-0.0014
	(0.0024)	(0.0016)	(0.0006)	(0.0014)
R^2	0.111	0.054	0.187	0.159
Ν	$720,\!254$	677, 323	$954,\!339$	690,201

Table 17: Subgroup Analyses - Differential Impacts of Rural and Urban Closures, Alternate "closure year"

Errors were not clustered by patient residential zip codes because the model did not converge with clustering. + p < 0.10, * p < 0.05

D Appendix: Other Sensitivity Analyses

Table 18: Differential Impacts of Rural and Urban Closures - Urban Patients Only

Variable	Marginal Effect (Std. Error)
Rural Treat Post	0.0044 +
	(0.0023)
Urban Treat Post	0.0006
	(0.0011)
Rural Treat	-0.0038
	(0.0031)
Urban Treat	-0.0012
	(0.0012)
R^2	0.148
Ν	$2,\!602,\!432$
Rural Treat Urban Treat R ² N	$\begin{array}{r} (0.0011) \\ -0.0038 \\ (0.0031) \\ -0.0012 \\ \hline \\ (0.0012) \\ \hline \\ \hline \\ 0.148 \\ \hline \\ 2,602,432 \end{array}$

Standard errors are reported in parentheses. Errors are clustered by patient residential zip codes. + p<0.10, * p<0.05

Table 19: Differential Impacts of Rural and Urban Closures - Sepsis, Stroke, AMI admissions combined

Marginal Effect (Std. Error)
0.0093*
(0.0029)
0.0021
(0.0016)
-0.0102*
(0.0033)
-0.0019
(0.0016)
0.104
1,867,244

Errors are clustered by patient residential zip codes.

+ p<0.10, * p<0.05

Table 20: Differential Impacts of Rural and Urban Closures - Length of Stay (LOS) as dependent variable

Variable	Marginal Effect (Std. Error)
Rural Treat Post	0.2259^{*}
	(0.0661)
Urban Treat Post	0.0248
	(0.0365)
Rural Treat	-0.0775
	(0.0826)
Urban Treat	-0.0128
	(0.0319)
R^2	0.157
N	2,736,902

Standard errors are reported in parentheses. Errors are clustered by patient residential zip codes. + p<0.10, * p<0.05

Table 21: Differential Impacts of Rural and Urban Closures - LOS as dependent variable; Sepsis, Stroke, AMI admissions combined

Marginal Effect (Std. Error)
0.3305^{*}
(0.0736)
0.0389
(0.0468)
-0.0655
(0.0849)
-0.0091
(0.0407)
0.154
1,867,503

Errors are clustered by patient residential zip codes.

+ p<0.10, * p<0.05

Table 22: Differential Impacts of Rural and Urban Closures - Inpatient mortality as dependent variable; controlling for growth in population density from 2000 to 2010

Variable	Marginal Effect (Std. Error)
Rural Treat Post	0.0047^{*}
	(0.0020)
Urban Treat Post	0.0010
	(0.0012)
Rural Treat	-0.0070*
	(0.0002)
Urban Treat	-0.0013
	(0.0012)
R^2	0.147
Ν	2,736,593

Standard errors are reported in parentheses.

Errors are clustered by patient residential zip codes. + p<0.10, * p<0.05

References

- AHA (2019). American hospital association (aha), landscape changes in u.s. hospitals. https://www.ahadataviewer.com/quickreport/. Accessed: 2019-05-21.
- Allen, J. (2017). Medicaid expansion and hospital closures: Examining hospital, county, and state effects in the wake of the affordable care act.
- Austin, P. C. (2011). An introduction to propensity score methods for reducing the effects of confounding in observational studies. *Multivariate behavioral research* 46(3), 399–424.
- Buchmueller, T. C., M. Jacobson, and C. Wold (2006). How far to the hospital?: The effect of hospital closures on access to care. *Journal of health economics* 25(4), 740–761.
- Carroll, Α. (2019).Healthcare triage: Rural hospital closures impact the of people. health of a lot https://theincidentaleconomist.com/wordpress/ healthcare-triage-rural-hospital-closures-impact-the-health-of-a-lot-of-people/ ?MessageRunDetailID=90939764&PostID=3665339. Accessed: 2019-05-21.
- Carroll, C. (2018). Impeding access or promoting efficiency? effects of rural hospital closure on the cost and quality of care, working paper. Job Market Paper.
- Chaou, C.-H., T.-F. Chiu, A. M.-F. Yen, C.-J. Ng, and H.-H. Chen (2016). Analyzing factors affecting emergency department length of stayusing a competing risk-accelerated failure time model. *Medicine* 95(14).
- CHCF (2015). California Health Care Almanac, California Health Care Foundation(CHCF), California Hospitals: An Evolving Environment. https://www.chcf.org/wp-content/uploads/ 2017/12/PDF-CaliforniaHospitals2015.pdf. Accessed: 2019-05-21.
- Dranove, D. (1998). Economies of scale in non-revenue producing cost centers: implications for hospital mergers. *Journal of Health Economics* 17(1), 69–83.
- Elixhauser, A., C. Steiner, D. R. Harris, and R. M. Coffey (1998). Comorbidity measures for use with administrative data. *Medical care*, 8–27.
- Escarce, J. J. and K. Kapur (2009). Do patients bypass rural hospitals?: Determinants of inpatient hospital choice in rural california. *Journal of health care for the poor and underserved* 20(3), 625–644.
- Frakt, A. (2019a). Jama forum: The rural hospital problem. https://newsatjama.jama.com/ 2019/05/01/jama-forum-the-rural-hospital-problem/. Accessed: 2019-05-21.
- Frakt, A. (2019b). A sense of alarm as rural hospitals keep closing. https://www.nytimes. com/2018/10/29/upshot/a-sense-of-alarm-as-rural-hospitals-keep-closing.html. Accessed: 2019-05-21.
- Friedman, A. B., D. D. Owen, and V. E. Perez (2016). Trends in hospital ed closures nationwide and across medicaid expansion, 2006-2013. The American journal of emergency medicine 34(7), 1262–1264.
- Gaynor, M., K. Ho, and R. J. Town (2015). The industrial organization of health-care markets. Journal of Economic Literature 53(2), 235–84.

- Hannan, E. L., C. Wu, T. J. Ryan, E. Bennett, A. T. Culliford, J. P. Gold, A. Hartman, O. W. Isom, R. H. Jones, B. McNeil, et al. (2003). Do hospitals and surgeons with higher coronary artery bypass graft surgery volumes still have lower risk-adjusted mortality rates? *Circulation 108*(7), 795–801.
- HASC (2010). Facility Closure History, Hospital Association of Southern California (HASC). http://www.hasc.org/sites/main/files/file-attachments/hasc_closure_ list_by_region_-_may_2010.pdf. Accessed: 2019-05-21.
- Henneman, P. L., B. H. Nathanson, H. Li, H. A. Smithline, F. S. Blank, J. P. Santoro, A. M. Maynard, D. A. Provost, and E. A. Henneman (2010). Emergency department patients who stay more than 6 hours contribute to crowding. *The Journal of emergency medicine* 39(1), 105–112.
- Hodgson, A., P. Roback, A. Hartman, E. Kelly, and Y. Li (2015). The financial impact of hospital closures on surrounding hospitals. *J Hosp Adm* 4(3), 25–34.
- Holmes, G. M., B. G. Kaufman, and G. H. Pink (2017). Predicting financial distress and closure in rural hospitals. *The Journal of Rural Health* 33(3), 239–249.
- Hsia, R. Y., H. K. Kanzaria, T. Srebotnjak, J. Maselli, C. McCulloch, and A. D. Auerbach (2012). Is emergency department closure resulting in increased distance to the nearest emergency department associated with increased inpatient mortality? *Annals of emergency medicine* 60(6), 707–715.
- Hsia, R. Y.-J. and Y.-C. Shen (2011). Rising closures of hospital trauma centers disproportionately burden vulnerable populations. *Health affairs* 30(10), 1912–1920.
- Joynt, K. E., P. Chatterjee, E. J. Orav, and A. K. Jha (2015). Hospital closures had no measurable impact on local hospitalization rates or mortality rates, 2003–11. *Health affairs* 34(5), 765–772.
- Kaufman, B. G., K. L. Reiter, G. H. Pink, and G. M. Holmes (2016). Medicaid expansion affects rural and urban hospitals differently. *Health Affairs* 35(9), 1665–1672.
- Kaufman, B. G., S. R. Thomas, R. K. Randolph, J. R. Perry, K. W. Thompson, G. M. Holmes, and G. H. Pink (2016). The rising rate of rural hospital closures. *The Journal of Rural Health* 32(1), 35–43.
- LA Times Data Desk (2019). California's Hospitals, Closed California Hospitals. http:// projects.latimes.com/hospitals/emergency-rooms/no/closed/list/. Accessed: 2019-05-21.
- Lindrooth, R. C., M. C. Perraillon, R. Y. Hardy, and G. J. Tung (2018). Understanding the relationship between medicaid expansions and hospital closures. *Health Affairs* 37(1), 111–120.
- Liu, C., T. Srebotnjak, and R. Y. Hsia (2014). California emergency department closures are associated with increased inpatient mortality at nearby hospitals. *Health Affairs* 33(8), 1323–1329.
- Manlove, J. and B. Whitacre (2017). Short-term economic impact of rural hospital closures. Technical report.
- McRae, A., I. Usman, D. Wang, G. Innes, E. Lang, B. Rowe, M. Schull, and R. Rosychuk (2017). Lo17: A comparative evaluation of ed crowding metrics and associations with patient mortality. *Canadian Journal of Emergency Medicine* 19(S1), S33–S33.

- Nicholas C Petris Center (2001). Californias closed hospitals, 1995-2000. Berkeley, Calif.: Nicholas C. Petris Center on Health Care Markets and Consumer Welfare, Berkeley School of Public Health, April.
- OIG (1995). Office of Inspector General (OIG), Department of Health and Human Services (DHHS), Report on Hospital Closure: 1995. https://oig.hhs.gov/oei/reports/oei-04-96-00060. pdf. Accessed: 2019-05-21.
- OIG (1996). Office of Inspector General (OIG), Department of Health and Human Services (DHHS), Report on Hospital Closure: 1996. https://oig.hhs.gov/oei/reports/oei-04-97-00110. pdf. Accessed: 2019-05-21.
- OIG (1997). Office of Inspector General (OIG), Department of Health and Human Services (DHHS), Report on Hospital Closure: 1997. https://oig.hhs.gov/oei/reports/oei-04-98-00200. pdf. Accessed: 2019-05-21.
- OIG (1998). Office of Inspector General (OIG), Department of Health and Human Services (DHHS), Report on Hospital Closure: 1998. https://oig.hhs.gov/oei/reports/oei-04-99-00330. pdf. Accessed: 2019-05-21.
- OIG (1999). Office of Inspector General (OIG), Department of Health and Human Services (DHHS), Report on Hospital Closure: 1999. https://oig.hhs.gov/oei/reports/oei-04-01-00020. pdf. Accessed: 2019-05-21.
- OIG (2000). Office of Inspector General (OIG), Department of Health and Human Services (DHHS), Report on Hospital Closure: 2000. https://oig.hhs.gov/oei/reports/oei-04-02-00010. pdf. Accessed: 2019-05-21.
- Pines, J. M., A. Prabhu, J. A. Hilton, J. E. Hollander, and E. M. Datner (2010). The effect of emergency department crowding on length of stay and medication treatment times in discharged patients with acute asthma. Academic Emergency Medicine 17(8), 834–839.
- Premkumar, D., D. Jones, and P. Orazem (2016). Hospital closure and hospital choice: How hospital quality and availability will affect rural residents.
- Rosenbach, M. L. and D. A. Dayhoff (1995). Access to care in rural america: impact of hospital closures. *Health Care Financing Review* 17(1), 15.
- Ruhm, C. J. (2000). Are recessions good for your health? The Quarterly journal of economics 115(2), 617–650.
- Ruhm, C. J. (2005). Healthy living in hard times. Journal of health economics 24(2), 341–363.
- Ruhm, C. J. (2007). A healthy economy can break your heart. Demography 44(4), 829–848.
- Song, L. and S. Saghafian (2019). Do hospital closures improve the efficiency and quality of other hospitals? *Available at SSRN 3318609*.
- Stagg, V. (2015). Elixhauser: Stata module to calculate elixhauser index of comorbidity.
- Stevenson, B. and J. Wolfers (2006). Bargaining in the shadow of the law: Divorce laws and family distress. The Quarterly Journal of Economics 121(1), 267–288.

- Strumpf, E. C., T. J. Charters, S. Harper, and A. Nandi (2017). Did the great recession affect mortality rates in the metropolitan united states? effects on mortality by age, gender and cause of death. Social Science & Medicine 189, 11–16.
- The Takeaway (2019). Hospitals in rural america are closing, with devastating consequences. https://www.wnycstudios.org/story/the-takeaway-2019-05-14. Accessed: 2019-05-21.
- Thomas, J. W., K. E. Guire, and G. G. Horvat (1997). Is patient length of stay related to quality of care? *Journal of Healthcare Management* 42(4), 489.
- Tran, H. V., D. Lessard, M. S. Tisminetzky, J. Yarzebski, E. A. Granillo, J. M. Gore, and R. Goldberg (2018). Trends in length of hospital stay and the impact on prognosis of early discharge after a first uncomplicated acute myocardial infarction. *The American journal of cardiology 121*(4), 397–402.
- Troske, S. and A. Davis (2019). Do hospital closures affect patient time in an ambulance?
- Wing, C., K. Simon, and R. A. Bello-Gomez (2018). Designing difference in difference studies: best practices for public health policy research. *Annual review of public health 39*.
- Wishner, J., P. Solleveld, R. Rudowitz, J. Paradise, L. Antonisse, et al. (2016). A look at rural hospital closures and implications for access to care: three case studies. *Kaiser Family Foundation [Internet]*.